

Emerging Technologies in K–12 Education: A Future HCI Research Agenda

MAARTEN VAN MECHELEN, RACHEL CHARLOTTE SMITH, MARIE-MONIQUE SCHAPER, MARIANA TAMASHIRO, KARL-EMIL BILSTRUP, MILLE LUNDING, MARIANNE GRAVES PETERSEN, and OLE SEJER IVERSEN,

Center for Computational Thinking and Design, Aarhus University

This systematic mapping review sheds light on how emerging technologies have been introduced and taught in various K–12 learning settings, particularly with regard to artificial intelligence (AI), machine learning (ML), the internet of things (IoT), augmented reality (AR), and virtual reality (VR). These technologies are rapidly being integrated into children's everyday lives, but their functions and implications are rarely understood due to their complex and distributed nature. The review provides a rigorous overview of the state of the art based on 107 records published across the fields of human-computer interaction, learning sciences, computing education, and child–computer interaction between 2010 and 2020. The findings show the urgent need on a global scale for inter- and transdisciplinary research that can integrate these dispersed contributions into a more coherent field of research and practice. The article presents nine discussion points for developing a shared agenda to mature the field. Based on the HCI community's expertise in human-centred approaches to technology and aspects of learning, we argue that the community is ideally positioned to take a leading role in the realisation of this future research agenda.

CCS Concepts: • General and reference \rightarrow Document types; Surveys and overviews • Social and professional topics \rightarrow Computing education; K-12 education;

Additional Key Words and Phrases: K–12 education, emerging technologies, computing education, computational literacy

ACM Reference format:

Maarten Van Mechelen, Rachel Charlotte Smith, Marie-Monique Schaper, Mariana Tamashiro, Karl-Emil Bilstrup, Mille Lunding, Marianne Graves Petersen, and Ole Sejer Iversen. 2023. Emerging Technologies in K–12 Education: A Future HCI Research Agenda. *ACM Trans. Comput.-Hum. Interact.* 30, 3, Article 47 (March 2023), 40 pages.

https://doi.org/10.1145/3569897

© 2023 Association for Computing Machinery.

1073-0516/2023/06-ART47 \$15.00

https://doi.org/10.1145/3569897

This work is part of a 5-year research project called **Computational Empowerment for Emerging Technologies in Education (CEED)**, exploring and building new practices of computational empowerment for emerging technologies in Danish secondary education through a cross-disciplinary approach between computer science, humanities, and engineering. The CEED-project is funded by VILLUM FONDEN under grant number 28831.

Authors' address: M. Van Mechelen, R. C. Smith, M.-M. Schaper, M. Tamashiro, K.-E. Bilstrup, M. Lunding, M. G. Petersen, and O. S. Iversen, Center for Computational Thinking and Design, Aarhus University; emails: maartenppvm@gmail.com, {rsmith, mmschaper, mariana}@cc.au.dk, {keb, milledsk, mgraves, oiversen}@cs.au.dk.

Permission to make digital or hard copies of all or part of this work for personal or classroom use is granted without fee provided that copies are not made or distributed for profit or commercial advantage and that copies bear this notice and the full citation on the first page. Copyrights for components of this work owned by others than ACM must be honored. Abstracting with credit is permitted. To copy otherwise, or republish, to post on servers or to redistribute to lists, requires prior specific permission and/or a fee. Request permissions from permissions@acm.org.

1 INTRODUCTION

This article provides an interdisciplinary account of how emerging technologies are being introduced and taught in K–12 settings, leading up to a future HCI research agenda. The use of emerging technologies such as **artificial intelligence** (**AI**), **machine learning** (**ML**), the **internet of things** (**IoT**), **augmented reality** (**AR**), and **virtual reality** (**VR**) has grown rapidly in children's everyday lives, but the general understanding of these technologies is still limited [84, 85]. This means that there is a significant imbalance between the far-reaching importance of emerging technologies and children's ability to comprehend these technologies and their consequences.

The urgent need on a global scale to educate K–12 students so they can critically and constructively engage in the digitisation of societies has brought with it an increased focus on K–12 teaching across a diverse range of research fields. In the past decade, **human-computer interaction** (**HCI**) research has produced novel technologies to support computing education [27, 55, 76], as well as overviews of the competences required for engaging with emerging technologies [85]. At the same time, the learning sciences are rapidly developing new didactics [123, 154], assessment practices [152], and specialised curricula [115, 136], which can support educators in introducing emerging technologies into K–12 teaching. Further, design research and especially **child–computer interaction** (**CCI**) research have studied children's perceptions of emerging technologies [143] and provided insightful design recommendations [82, 121, 164] as well as approaches to teaching about emerging technologies [2, 41].

Policymakers at national and international levels as well as non-governmental organisations have called for action to provide children with knowledge, skills, and competences in relation to digital technology, digitisation and, more recently, emerging technologies. Initiatives include, among others, the Digital Education Action Plan [47] and Informatics for All coalition [69] in Europe, and in the United States, the National Artificial Intelligence Initiative [140], with one of six strategic pillars focused on Training and Education. UNICEF (United Nations Children's Fund), in turn, has developed guidelines for "AI and child rights policy" together with a number of renowned international experts [139]. Their premise is that "the rights of children, as current users of AI-enabled systems and the future inhabitants of a more AI-saturated world, must be a critical consideration in AI development" [139]. It is argued that integrating the benefits of novel technologies, computational thinking, digital literacy, and maker education will enable children to actively participate in a society with increasing human-computer interaction [19–21, 101].

In response to these urgent calls and to identify progress to date across academic research, we conducted a systematic mapping review of the state of the art of emerging technologies in K–12 education, based on 107 records published between 2010 and 2020. The review covers the five central topics of (1) target groups and teacher roles (Section 4.2), (2) learning objectives and curricular implementation (Section 4.3), (3) educational frameworks and practices (Section 4.4), (4) technology and other learning tools (Section 4.5), and (5) empirical evaluation and student assessment (Section 4.6). It identifies trajectories in the current literature across fields, and provides a strong starting point for developing a shared research agenda in emerging technologies in K–12 education (Section 5).

The HCI community is ideally positioned both to take a leading role in the realisation of this inter- and transdisciplinary agenda and to act as a mediator between neighbouring and subfields for diverse reasons. Firstly, it has a proven track record of human-centred approaches and commitment to understanding people's entanglement with technology [11, 49]. For decades, HCI researchers have investigated how people learn to use and adapt technology tools, with "designing for learnability" as a key HCI design principle [160]. The HCI community's interest in aspects of learning has also been manifested in the design of educational technologies (ed-tech), providing a second reason for why the community has a pivotal role in the realisation of this agenda.

Designing for learning (e.g., MOOCs, intelligent tutoring systems, VR learning tools) has specific design and interaction issues that go beyond more generally applicable design knowledge, witnessed by the increased engagement of learning scientists with the HCI community. This has, in turn, led to the creation of the newly established Learning, Education and Families subcommittee with 179 articles submitted for CHI 2020 [160]. A third reason is the community's ongoing commitment for advancing UX and HCI education by addressing the changing needs of students and professionals in computing and interaction design [156]. This has brought forth several curricular innovations in the domain of technology education (tech-ed) such as the integrated studio approach for teaching HCI to undergraduates [153]. By drawing on this extensive expertise in human-centred approaches, ed-tech and tech-ed, and bringing in researchers from neighbouring fields, the HCI community can act as a catalyst for change and create a lasting impact on children's agency in an increasingly digitalised society – an impact that reaches beyond research into policy and education, as suggested by Bødker and Kyng [12].

The article is organised as follows. Section 2 defines the concept of emerging technology and discusses considerations for K–12 education. Section 3 describes the review method and explains in detail how the literature was collected, analysed, and synthesised. Section 4 presents the results, including a description of the dataset and findings for each of the five central topics of interest. Section 5 discusses the main findings, leading up to nine discussion points for developing a shared agenda for future research. Finally, Section 6 summarises the main conclusions of the systematic mapping review.

2 EMERGING TECHNOLOGIES: DEFINITIONS AND CONSIDERATIONS

Emerging technologies have only recently emerged as a topic in education [14, 19, 21, 25–27], and still receive only a fraction of the attention devoted to the digitisation and digital technology more generally [83]. In this section we discuss the unique characteristics of emerging technologies and argue why addressing these technologies – AI, ML, IoT, AR, and VR in particular – is critical for K–12 education.

Many definitions of emerging technology have been suggested [30, 31, 130, 131]. An early definition is provided by Martin [89], who emphasises its transformative impact as a key characteristic, describing it as "a technology the exploitation of which will yield benefits for a wide range of sectors of the economy and/or society" (p. 165). In addition to the potential impact and the transformative nature of emerging technologies, Day and Schoemaker [32] emphasise its origins in radical innovation and/or technology convergence. They describe an emerging technology as "a science-based innovation that has the potential to create a new industry or transform existing ones. It includes discontinuous innovations derived from radical innovations (...) as well as more evolutionary technologies formed by the convergence of previously separate research streams" (p. 30). Boon and Moons [13], in turn, emphasise a further characteristic, uncertainty, due to the fact that these technologies are still in the early development stage. This means that "several aspects, such as the characteristics of the technology and its context of use or the configuration of the actor network and their related roles are still uncertain and nonspecific" (p. 1915).

Building on this work, Rotolo et al. [114] foreground five characteristics of emerging technologies: their radical novelty, their relatively fast growth, a certain degree of coherence and momentum, their significant impact on specific domains or on society more broadly, and uncertainty and ambiguity about possible outcomes and uses. They define an emerging technology as

a relatively fast-growing and radically novel technology characterised by a certain degree of coherence persisting over time and with the potential to exert a considerable impact on the socioeconomic domain(s) which is observed in terms of the The internet of things (IoT) describes physical objects or 'things' that are embedded with sensors, processing ability, software, and other technologies that connect and exchange data with other devices and systems over the internet or other communications networks.

Artificial intelligence (AI) is intelligence demonstrated or simulated by technological means, as opposed to natural intelligence. It allows computers and robots to do tasks that are usually done by humans because they require human intelligence and discernment.

Machine learning (ML) is a type of AI that allows software applications to become more accurate at predicting outcomes without being explicitly programmed to do so. ML algorithms use historical data as input to detect patterns and predict new output values.

Virtual reality (VR) is a computer-generated environment with scenes and objects that appear to be real, making the user feel they are immersed in their surroundings. This environment is typically perceived through a device known as a VR headset or helmet.

Augmented reality (AR) is an interactive experience of a real-world environment where objects are enhanced by computer-generated perceptual information, sometimes across multiple sensory modalities. Unlike VR, which creates its own computer-generated environment, AR adds to the existing world as it is.

Fig. 1. Generic descriptions of the emerging technologies are under consideration in this article.

composition of actors, institutions and the patterns of interactions among those, along with the associated knowledge production processes. Its most prominent impact, however, lies in the future and so the emergence phase is still somewhat uncertain and ambiguous. (p. 1,828)

This definition frames emerging technology as radically novel in using a different basic principle to fulfil a given function than what was previously used to achieve a similar purpose, thereby changing the status quo [114]. However, the attribute of novelty is, to a large extent, context-specific. A technology can be considered "old" and no longer emerging in one domain (e.g., the commercial or business sector) while still being considered emerging in another (e.g., the education sector) where the potential growth and socioeconomic impact of a technology are still largely untapped [94]. Moreover, what is considered an emerging technology may differ among actors, depending on how they conceive the expected growth and impact [114].

To be labelled as emerging, though, a novel technology requires a certain coherence compared to others that are still in flux and lacking in established momentum and/or a relatively fixed identity (lacking, e.g., an agreed-upon label and abbreviation) [114]. This does not mean that emerging technologies have yet reached technology maturity, as they tend to have incomplete specifications and nonuniform standards. Low technology maturity means that, especially at the beginning, only a few ethical and societal issues are observable. As time passes, however, and as an emerging technology is used in different applications, the number of new and unanticipated ethical and societal concerns it throws into relief gradually increases. At the same time, the technology matures, resulting in lowered costs and uncertainty levels as possible uses, outcomes, and associated meanings become apparent [64].

Emerging technology is a term that applies to a broad range of application domains, including but not limited to agriculture (e.g., cultured meat), medicine (e.g., RNA vaccine technology), construction (e.g., four-dimensional printing), finance (e.g., digital currency), robotics (e.g., powered exoskeletons), and materials science (e.g., bioplastic) [37, 38, 72]. In this article, we confine ourselves to the domain of IT and communication, and in particular to those emerging technologies that children are increasingly exposed to in their everyday lives in school, family, and leisure time. These include the IoT, VR, AR, AI, and ML (see Figure 1). Examples of such exposure are AR filters in Snapchat and TikTok, ML-powered recommendations in YouTube and Netflix, popular VR games such as AstroBot or Beat Saber, educational applications including Froggipedia

and Mondly, smart home assistants like Alexa and Google Assistant, and the "Internet of Toys" – Hello Barbie, Little Bits R2D2, and other connected toys [90]. These technologies are rapidly becoming integrated into aspects of everyday life, but are rarely understood due to their complex and distributed nature, which allows little transparency into their functions and implications. Children's use of and exposure to these emerging technologies, often unknowingly, has increased in recent years, but the full impact of these technologies on children's lives is yet to be realised [6, 16, 127, 161]. It is therefore important that children move beyond passive consumption to develop the competences and critical understanding they will need to maximise the benefits and opportunities that these technologies offer, now and in the future, while limiting exposure to risks and potential harm [19, 24, 84].

Although the selected emerging technologies differ in their complexity and in their prevalence in children's lives, they are rapidly becoming ubiquitous and, to some degree, relate to topics of growing interest in K–12 computing education such as data literacy and cybersecurity. The selection furthermore captures the characteristics and potential impacts of emerging technologies that make them so challenging and ambiguous to integrate in learning processes for coming generations.

3 METHOD

The aim of this article is to provide a state-of-the-art review of emerging technologies – and of AI, ML, AR, VR, and IoT in particular – in K–12 education (see the typology of reviews by [62]). We conducted a critical analysis and synthesis of peer-reviewed archival and non-archival records published in English between 2010 and 2020 that focus on introducing or teaching emerging technologies in different educational settings. The main contribution of this state-of-the-art review is a well-grounded agenda for further research to improve the knowledge base and advance this emerging field [128, 135]. To this end, as suggested by [110], we used a systematic approach combining electronic and manual searching to identify and select relevant literature from a broad range of fields.

The review has five main focus areas. It examines (1) who is the target group of the learning activities, and what are the roles of teachers and other actors in preparing and facilitating these activities, (2) what is being taught about emerging technologies and for what reasons, (3) how it is being taught and what theoretical frameworks are relied on, (4) what technology and other tools are used to this end, and finally (5), how practices and tools are evaluated and students' learning assessed. In the following section we explain in detail how we collected, analysed, and synthesised literature, resulting in a substantial agenda for future research in emerging technologies in K–12 education.

3.1 Data Collection

To ensure transparency and to limit bias in the selection of literature, we followed the PRISMA-ScR guidelines and flow diagram (see Figure 3) [106, 138]. We first conducted an electronic search, then screened and assessed records for eligibility, and finally searched the included records for relevant citations.

3.1.1 Step 1: Electronic Search Query. An electronic search was conducted in the Scopus database (+80 million records), which integrates the ACM Digital Library, IEEE, and Elsevier databases, among others (see Figure 2). Only records with the exact or stemmed words of the inclusion criteria in either the title, abstract, or keywords were included. The inclusion criteria can be divided into three groups. The first of the three relates to the target group (including terms such as children, kids, youngsters, K–12, students, or pupils), the second to the educational context (terms such as education, learning, teaching, school, classroom, or literacy), and the third to the type of technology (terms such as emerging technology, AI, ML, AR, VR, or IoT). Our reasons for (TITLE-ABS-KEY (K12* OR child* OR kid* OR teen* OR pupil* OR student* OR youngster*)
AND TITLE-ABS-KEY (learn* OR teach* OR educat* OR literacy OR literate OR school* OR classroom*)
AND TITLE-ABS-KEY ("emerging technolog*" OR "artificial intelligence" OR ai OR "machine learning" OR ml OR "augmented reality" OR ar
OR vr OR "internet of things" OR iot)) AND NOT (universit* OR college OR "higher education")
AND (LIMIT-TO (DOCTYPE, "cp") OR LIMIT-TO (DOCTYPE, "ar") OR LIMIT-TO (DOCTYPE, "ch") OR LIMIT-TO (DOCTYPE, "bk") OR
LIMIT-TO (DOCTYPE, "re")) AND (EXCLUDE (SUBJAREA, "MEDI") OR EXCLUDE (SUBJAREA, "HEAL") OR EXCLUDE (SUBJAREA,
"NEUR") OR EXCLUDE (SUBJAREA, "NURS") OR EXCLUDE (SUBJAREA, "PHAR") OR EXCLUDE (SUBJAREA, "IMMU") OR
EXCLUDE (SUBJAREA, "DENT") OR EXCLUDE (SUBJAREA, "VETE")) AND (LIMIT-TO (PUBYEAR, 2020) OR LIMIT-TO (PUBYEAR, 2019) OR LIMIT-TO (PUBYEAR, 2014) OR LIMIT-TO (PUBYEAR, 2017) OR LIMIT-TO (PUBYEAR, 2016) OR LIMIT-TO (PUBYEAR, 2015) OR LIMIT-TO (PUBYEAR, 2014) OR LIMIT-TO (PUBYEAR, 2013) OR LIMIT-TO (PUBYEAR, 2012) OR LIMIT-TO (PUBYEAR, 2011) OR LIMIT-TO (PUBYEAR, 2010))

Fig. 2. Search query used to identify records in the Scopus database (incl. IEEE, ACM DL, and so on.).

compiling the search terms as above were twofold: first, so many children are encountering these technologies on a very frequent basis that it is critical that they learn about them and their potential impact from an early age; and second, we wanted to limit the corpus of included records to a manageable size.

To further limit the scope, only records published in English between 2010 and 2020, and in the form of peer-reviewed articles, conference proceedings, reviews, book chapters, and books were included. The search was limited in time because emerging technologies in K–12 education are a recent phenomenon, and the peer-review criteria were used to ensure a minimum academic quality for the included records. To focus on K–12 education, any records using the words "university," "college" or "higher education" in the title, abstract, or keywords were excluded.

3.1.2 Step 2: Screening and Assessing for Eligibility. Using this search query (see Figure 2), 1,873 records were retrieved from the Scopus database by April 4, 2021. Based on an initial screening of the title and abstract by the first author, 1,357 records were excluded because they were duplicates, were not published in English, or did not focus on education. The remaining 550 records were screened and assessed for eligibility by the first author and an additional researcher based on the title, abstract, and, if needed, the full text. The researchers worked independently and flagged any doubtful cases. The results were compared, followed by a discussion of all doubt cases and disagreements between the two coders (n = 120). For the majority of records, a final decision was reached this way (n = 75). In other cases, an additional check of the target group was required (n = 16), or the record had to be screened and assessed by a third researcher (n = 29). In this process, 476 records were excluded because either the target group did not correspond with the inclusion criteria (n = 104), the focus was on educational technology (e.g., using AR to teach geography) rather than technology education (n = 312), or the topic was considered irrelevant (e.g., no reference to emerging technologies, education mentioned but not the main focus) (n = 58). From the remaining 76 records, two could not be retrieved. This left a total of 74 records, which were downloaded and organised in folders per publication year.

3.1.3 Step 3: Citation Searching. All 74 included records were scanned for relevant citations by the first author, using the same inclusion and exclusion criteria as for the electronic search. This "snowballing" process resulted in 33 additional references, which (as in the preceding stage) were screened and assessed for eligibility by the first author and an additional researcher based on the title, abstract, and, if needed, the full text. One record was removed because the body text was not in English, but there were no further disagreements between the two researchers. The remaining 32 records were downloaded and sorted in the folders per publication year. This resulted in a total



PRISMA 2020 flow diagram for new systematic reviews which included searches of databases, registers and other sources

Fig. 3. Flow diagram based on the PRISMA-ScR guidelines [106, 138].

Table 1. Data Descriptors and Leading Questions were used to index all 107 Included Records

Data descriptor	Leading question
Publication year	When was the publication published?
Publication venue	Where is the publication published (name of the conference, journal, book series, and so on)?
Publication type	What type of publication is it (archival or non-archival conference proceeding, journal article, or book chapter)?
Geographical distribution	In which country is the first author's institute located?

of 107 records included for review. 65 of these records report original empirical data (for a detailed description of the dataset, see Section 4) (see Figure 3).

3.2 Data Analysis

3.2.1 Step 4: Deductive Analysis based on Predefined Categories. To analyse the records, a spreadsheet was created with separate columns for descriptive information including the title, author information, abstract, year of publication, publication venue, publication type, and geographical distribution (see Table 1). Additional columns were created for the deductive analysis categories, which were defined based on the research question (RQ):

What is the state of the art of teaching emerging technologies in diverse K-12learning settings, with particular regard to AI, ML, AR, VR, and the IoT?

The five deductive analysis categories are "target group, teacher roles and other actors," "learning objectives and implementation," "educational frameworks and practices," "technology and other tools for learning," and "evaluation and/or assessment" (see Table 2). A final set of columns was created to summarise the overall relevance of the record for the review, and to list any relevant references or other resources.

First, each record was read carefully by the first author and descriptive information was filled out in the spreadsheet. Then, relevant information for each of the five main categories was copied into the dedicated cells with reference to the page number. In some cases, this required several readings of the same record. Once this step was completed, a short summary was added for each main

Main categories or parent codes	Leading question per category	Inductive child codes for each category
Target group, role of teachers and any other actors (section 4.2)	Who is the target group of the learning activities and tools, and what is the role of teachers and other actors (backstage and/or frontstage)?	Age-range target group Number of participants in study Role of researchers Role of teachers (backstage and/or frontstage): Role of other actors
Learning objectives and (curricular) implementation (section 4.3)	What are the learning objectives and to what extent are these implemented in or across (new) curricula?	Standalone activities or curricular integration Higher-order objective (career vs. literacy) Progressive objectives Learning objectives: - Technology concepts - Technology skills - Societal/ethical implications - Design knowledge/skills - Other knowledge/skills - Attitudes and mindsets Prior knowledge/skills requirements
Educational frameworks and practices (section 4.4)	How is children's learning supported, and to what extent are these practices grounded in the corpus of learning theory?	Theoretical learning framework Pedagogical strategies: - Hands-on - Collaborative - Theory-driven - Adaptability - Other strategies Format and duration of activities Formal, informal, or non-formal learning
Technology and other tools for learning (section 4.5)	Which technology and other (unplugged) learning tools are proposed/used in the activities?	Type of emerging technology: AI/ML, IoT, AR/VR Off-the-shelf technology tools New and new combinations of technology tools Unplugged tools Other characteristics of tools
Empirical evaluation and assessment of children's learning (section 4.6)	How are the learning activities and tools empirically evaluated, and/or children's learning assessed?	Evaluation of learning tool(s) Evaluation of learning activities Assessment of learning: – Summative assessment – Formative assessment – Constructive alignment Type of data collection Type of findings

Table 2. The five main Deductive Categories, the Leading Questions for each Category,and data-driven Child Codes

category in these cells, as well as a note about the overall relevance of the record for the review. Finally, interesting references and other relevant resources were added to the last column of the spreadsheet. This process was repeated for all 107 records, after which a second coder skimmed each record and verified the deductive analysis conducted by the first author. Three co-authors took the role of second coder, thereby focusing on records matching their particular expertise in either AI and ML, AR and VR, or IoT. Any inaccuracies and disagreements were flagged and discussed in plenum until an agreement was reached.

3.2.2 Step 5: Inductive Analysis within and Across Categories. The next step involved an inductive analysis for each of the five main deductive categories (or parent codes) across all 107 records. First, we identified data-driven child codes by going through the tabular data per category. These child codes were interpreted from the data and once an initial set of codes was established for each of the five categories, we duplicated the respective columns in the spreadsheet and conducted a more fine-grained analysis by restructuring the tabular data in the duplicated columns according to each child code. This was a highly iterative process of going back and forth between different records, somewhat akin to "open coding" [132], which required a few child codes to be omitted or altered along the way. The final code scheme is presented in Table 2.

Once we had coded all the data extracted from the included records, we sought for possible patterns within and across child codes in a process akin to "axial coding" [132]. As with the previous step, these patterns were interpreted from the data. Examples of such patterns include the co-occurrence of pedagogical strategies coded as "hands-on" and "collaborative," and the strong link between the child codes "societal and ethical implications" and "AI/ML" (see Appendix).

Next, we summarised the findings per the main category in a separate document that was verified by the same researchers who had acted as second coder in the previous step. We then compared findings across categories, in a process akin to "selective coding" [132], and collaboratively identified major implications of the review and key topics for a future research agenda. Finally, we reported both the detailed findings, which we illustrated with examples from the reviewed records (see Section 4), and the agenda topics (see Section 5).

3.3 Limitations

As with any study, this review has a few limitations. First, we used only the Scopus database to identify and select relevant literature. Although Scopus integrates multiple relevant databases (e.g., Elsevier, ACM DL, IEEE), a crosscheck with Web of Science or other databases might have been productive. As an alternative, we combined electronic search with citation screening to identify any records that we might have missed. Another limitation is that we included both archival and non-archival records (but not grey literature). This was done to gain a good overview of current research on emerging technologies in K-12 education, while ensuring academic quality by including only peer-reviewed records. A third limitation is that we limited the scope of the review to the emerging technologies with which we believe children are already interacting on a frequent basis: that is, artificial intelligence and machine learning, augmented and virtual reality, and the internet of things. In future work, the review could be extended to include additional emerging technologies (e.g., humanoid and social robotics, e-textiles, holographic displays) and/or topics of growing interest in K-12 computing education (e.g., data literacy, cybersecurity, quantum computing) to provide an even more comprehensive overview. Related to this, the review does not include records published after 2020, although emerging technologies and research on education surrounding these topics have continued to expand rapidly. A fourth and final limitation is that, due to pragmatic considerations, the deductive and inductive analysis was conducted by one researcher (i.e., the first author) and only verified afterwards by three co-authors. This stands in contrast with the selection process, in which two researchers independently screened and assessed all records.

4 RESULTS

The results section is structured by the main deductive categories used to analyse the 107 included records. First, we describe the dataset, including the types of records, publication venues, publication year, geographical distribution, and the emerging technology focused on (Section 4.1). Next, we discuss the target groups mentioned in the records, as well as the role of teachers and other



Fig. 4. Distribution of included records (n = 107) per technology and publication year.

actors involved in the preparation and/or facilitation of learning activities (Section 4.2). Then, we move on to discussing the learning objectives - both high- and low-level - and the ways in which learning objectives are integrated into existing or new curricula, if at all (Section 4.3). Following this, we dive into the theoretical learning frameworks and practices reported in the records. Here we also give an overview of preferred pedagogical strategies (Section 4.4). From there, we move on to discussing technology and other tools developed or used by researchers to support educational practices (Section 4.5). Finally, we look at how these practices and tools were evaluated, and if and how students' learning was assessed (Section 4.6). Important to note is that we grouped AI and ML as well as AR and VR in the presentation of the findings. In addition, we only distinguished between the three groups of emerging technologies when the analysis of the data revealed clear differences in the ways in which these technologies are taught in K-12 education.

Description of the Dataset 4.1

This section describes the dataset, starting with the distribution of records across publication year and type of emerging technology. As Figure 4 clearly shows, interest in teaching emerging technologies such as AI, ML, IoT, AR, and VR in K-12 education has increased. From the 107 records included in the review, 57 primarily focus on AI/ML, 28 on IoT, and 27 on AR/VR. Only five articles focus on more than one of these technologies, of which three combine IoT with AR/VR and two IoT with AI/ML. The vast majority of articles (82 out of 107) were published in the last three years, with 27 records in 2020, 36 in 2019, and 19 in 2018. Especially for records that focus on teaching AI and ML (57 in total), there has been a sharp increase in the past two years, with 18 records published in 2020 and 20 in 2019.

The 107 records were published in 78 different venues, including 58 conferences, 17 journals, and three books. Within this broad range of venues, the following groups can be distinguished: HCI and IxD venues (13 venues for 27 records), learning sciences (four venues for 26 records), computing and/or engineering education (19 venues for 26 records), learning technologies or EdTech (16 venues for 20 records), computing and/or engineering (14 venues for 14 records), intelligent systems (10 venues for 12 records) and miscellaneous (two venues for two records). Put differently,



Fig. 5. Geographical distribution of included records across continents (n = 107).

24 venues have a predominantly technical focus, 23 venues focus on (technology) education, 16 on educational technology, and 13 on HCI and IxD.

Among individual venues, Interaction Design & Children Conference (IDC) is the most represented (9 records), followed by CHI (Conference on Human Factors in Computing Systems) (6 records), and Innovation and Technology in Computer Science Education Conference(ITiCSE) (4 records). Runners-up are Frontiers in Education Conference (FIE), Global Engineering Education Conference (EDUCON), International Conference on Teaching, Assessment, and Learning for Engineering (TALE), European Conference on Technology Enhanced Learning (ECTEL), and AAAI (Conference on Artificial Intelligence) (three records each), and the International Journal of Child–Computer Interaction, Conference on Informatics in Schools: Situation, Evolution, and Perspectives (ISSEP), Symposium on Educational Advances in Artificial Intelligence (EAAI), and T4E (International Conference on Technology for Education) (two records each). All other venues are represented once only. With regard to the publication type, the selection includes 56 (full) papers, 16 journal articles, three book chapters, and 32 non-archival publications (e.g., magazine article, work in progress, demo).

With regard to geographical distribution (i.e., the country in which the first author's home institute is located), 30 countries are represented in the selection. The United States is represented by approximately one-fifth of the records (21 in total), followed by Spain (11 records) and Greece (9 records). China, Germany, and India each have six records in the selection, Finland and Israel five, Portugal and the UK four, and Brazil and Italy three. The remaining countries are represented by either one or two records. As for distribution across continents, Europe is represented by half of the records (53 in total), Asia by 25, North America by 23, South America by 4, Australia by 2, and Africa by none (see Figure 5). Important to note here is that we selected only English records for this review (see Section 3 for more details about inclusion and exclusion criteria).

In sum, the vast majority of records included for review were published in the last three years, and records focused on AI and ML especially have sharply increased. When looking at the venues in which these records were published, we observe a large diversity of technical, educational,



LEVELS OF EDUCATION

Fig. 6. Distribution of included records (n = 107) per aggregated level of education: preschool, primary education (grades 1–6) and/or secondary education (grades 7–12).

and HCI/IxD-oriented conferences and journals. Less diversity was found in the geographical distribution of the institutions of the first authors, showing a clear overrepresentation of Western countries.

4.2 Target Group and Roles for Teachers and other Actors

In this section, we look at the primary target groups for the learning activities, and at any other actors playing an active role in the development (i.e., backstage work) and facilitation of these activities, most notably researchers and teachers, as presented in the 107 reviewed records.

The most common target group for learning activities teaching about emerging technologies is students in secondary education, in particular, grades 8 to 10 (ages 13–16) (e.g., [57, 81, 163]). Lower primary education, and especially preschool, are less represented (e.g., [70, 75, 158]) (see Figure 6). Of the 107 included records, eight use generic terms such as "K12" or "secondary education," and 17 do not mention a specific grade or age range. All other records provide detailed information about the target group, which often covers several grades. Very few reports focus on specific groups of students, such as girls or other underrepresented groups in computing education [79]. On the other hand, gender balance seems to be an explicit aim in several studies when recruiting participants (e.g., [28, 54, 120, 163]).

Of the 107 included records, 65 reports original empirical work, with an average of 34 students and a mean value of 25 students per study. Thus, most studies seem to aim for "depth" with fewer students rather than "breadth" with many respondents. Notable exceptions include one study with 150 students [129] and another that impacted over 3,000 students by rolling out a STEM and IoT programme in 18 schools [79]. In contrast, two studies involved as few as three students [44, 58]. Important to note is that we refer to "a study" here also as a combination of multiple studies (e.g., a presurvey, pilot study, and main study) reported in a single article.

Teachers are frequently referred to as important actors in learning activities. Teachers' roles, and the ways in which they collaborate with researchers, are specified in fewer than one-third of all records (33 in total). Although researchers take the lead in preparing learning activities (i.e., the

backstage work), in 12 records teachers are actively involved in this stage, either by co-creating learning content and activities with researchers or simply by providing feedback on initial work prepared by researchers (e.g., [78, 97, 115, 134]). The co-creating option gives teachers the most impact on and ownership of the outcomes of this preparation stage. Teachers are also involved in the next stage, that is, facilitating learning activities in formal and non-formal learning settings. In 14 records, teachers either co-facilitate the activities with the leading researchers (e.g., [123, 126, 167]) or facilitate the activities alone while researchers attend the sessions passively, for instance, by taking notes and collecting research data (e.g., [14, 46, 58]).

Only in a few cases (five records) are teachers actively involved both in backstage work to prepare an intervention and in facilitating the resulting learning activities. Charlton and Avramides [23], for instance, actively involved teachers in constructing knowledge and experimenting with ideas on how an IoT system could be used for collaborative, problem-based, and multidisciplinary STEM education. Heinze et al. [65] in turn, report on a three-year-long collaboration between an AI researcher and two local teachers to develop a K–6 AI curriculum as part of the Scientistsin-Schools program in Australia. The learning objectives, content, and activities were developed collaboratively and tried out by the teachers across subjects and in multiple iterations.

In just two records, teachers do not partake in backstage work or facilitation of interventions, but instead participate in in-service and professional development programmes set up by researchers [79, 148]. The aim of these programmes is to train teachers to integrate emerging technologies in K–12 education, something that is deemed important for scaling and sustaining research-led initiatives on technology education. On that account, Vazhayil et al. [148] developed a course for teachers on how to introduce AI to middle and high school students. 34 teachers with different educational backgrounds and levels of experience participated in the course. They learned theoretical concepts of AI and the various stages of an AI project cycle, and afterwards applied this knowledge in CS subjects in their respective schools.

Besides K–12 students (i.e., the target group) and teachers, only a few records (10 in total) mention active involvement by other actors such as parents [146], additional researchers and/or industry partners [149], education experts [151] and policymakers [162]. The role of these actors is diverse, ranging from the co-facilitation of learning activities to providing input for the development of these activities.

In sum, the primary target group for learning activities for emerging technologies is students in secondary education, in particular grades 8 to 10. Very few records specifically address very young and underrepresented target groups. On average, 34 students participate in a single study, meaning that "depth" with fewer respondents is a preferred research strategy. Including a broader range of students from early years throughout secondary education, and complementing small with larger cohorts of student participants, would offer great potential in terms of outreach and diversity. As for teachers, we found their role in learning activities for emerging technologies to be fairly limited: fewer than a third of all records involved teachers in the development, facilitation, or evaluation of learning activities. A further limiting factor on long-term impact was the scarcity of in-service and professional development programmes, which are important for scaling and sustaining research-led initiatives. These findings are consistent across all three types of emerging technologies.

4.3 Learning Outcomes and (Curricular) Implementation

This section zooms in on the learning objectives associated with emerging technologies, the implementation of those objectives, and additional overlapping and interconnected aspects. In the analysis, we first looked for information about the higher-order objective for teaching emerging technologies in K–12 education. Next, we looked at the different types of concrete learning objectives: these formed a disparate list, ranging from technical understanding and skills, design skills,

to societal and ethical implications, attitudes, and other objectives including developing STEM and transversal skills. We then analysed whether learning objectives were integrated with existing or new curricula, their level of detail or abstraction and whether they account for a range of progression levels, and lastly, the extent to which prior knowledge and skills are required to participate in the learning activities targeting these objectives.

A higher-order objective for the authors' belief that children should be taught about emerging technologies – AI, ML, AR, VR, and IoT in particular – is identified in fewer than half the records (48 out of 107). In the other half, such a higher-order objective is either missing or not explicitly communicated. For those that do mention a higher-order objective, a distinction can be made between records foregrounding a career perspective (20 records), often in STEM disciplines (e.g., [3, 60, 63]), and those that advocate a broad literacy perspective (28 records), most often with regard to AI and ML (e.g., [37, 72, 80, 88]).

In the literacy perspective, education about emerging technologies is considered relevant for all children, regardless of future career trajectories, as with the core subjects of maths, reading, and writing. In this perspective, developing a critical understanding of emerging technologies and the operational skills to work with them will in the near future be a precondition to full participation in society. A good example is provided by Druga et al. [42], whose aim is to develop children's AI literacy through physical tinkering and learning activities using smart toys and agents. The authors argue that it is important for children in contemporary society to understand how machines perceive and model the world as they grow up with these technologies. Tissenbaum et al. [133], in turn, argue that providing low-barrier means for all types of students to design and implement IoT solutions to problems that have personal relevance to them can help them develop computational identities or "the ability to create meaningful change using computing and recognising one's place in the large computing community," as well as their sense of digital empowerment or "opportunities to put that identity into action." These notions of identity and empowerment clearly point to a broad literacy perspective, even if the term itself is not used [133].

In addition to higher-order goals, we assessed whether learning activities were integrated in or across existing curricula, or presented as new curricula altogether. We found that almost half of the records (48 out of 107) lack such integration and present standalone learning activities (see Figure 7). This means that researchers are conducting activities in formal or non-formal learning contexts, during or after regular school hours, but without integrating the objectives they are introducing with established ones (e.g., [17, 73, 82, 86]).

If not presented as standalone activities, curricular integration is quite common in STEM (19 records, of which ten focus on IoT: e.g., [96, 119]) and **computer science** (**CS**) education (9 records: e.g., [75, 149]) (see Figure 7). Along these lines, Ota et al. [105] combined general STEM objectives, especially in relation to mathematics, with IoT-specific objectives. They developed a 15-hour STEM course with IoT learning materials and modules in which high school students create prototypes that solve real-world problems of their own choosing. Through this process, students learn how to collect and analyse sensor data with probability statistics (i.e., mathematics).

An alternative approach to integrating learning activities in a single school subject is a crosscurricular approach, pushing against traditional subject boundaries (10 records, of which seven focus on AI/ML: e.g., [23, 142]) (see Figure 7). A good but rare example of a cross-curricular approach is provided by Chow [26], who engaged high school students in a seven-month-long project that ran across different subjects. Students collaborated in small groups to create a VR model of their school campus. This required them to learn a range of skills, many of which were not covered in their regular school subjects. Among other things, students learned advanced programming, 3D modelling, project-management skills, synthesising literature, heritage science as applied to the history of their school campus, and graphic design.



Fig. 7. Distribution of records (n = 107) per technology and integration of activities in/across curricula.

A further approach, taken by six records, is to propose a dedicated curriculum, covering one or multiple grades, for introducing emerging technologies in K–12 education (e.g., [115, 162]) (see Figure 7). All six of these records focus on AI/ML. Gong et al. [59] for instance, present an "AI education system" for primary and high school students including different cognitive and practice-based objectives, hardware and software tools, and concrete cases that can be used in a modular way depending on students' level and grade.

With regard to the degree of detail of the learning objectives, half of the records (57 in total) provide only high-level descriptions with little operational detail. In contrast, 12 records provide very detailed learning objectives (e.g., [48, 85]). Only a few records (9 in total) suggest progressive learning objectives, such as specifying objectives per grade for a dedicated technology curriculum (e.g., [33, 65]). Often cited in this regard is Touretzky et al. [137], who present five big ideas about AI and detail what students in different grades should be able to do and know in relation to each of them.

Although providing low barriers to entry seems to be the norm for learning activities about emerging technologies, in one-fifth of the reviewed records (22 in total) a certain degree of prior skills and knowledge is required to participate in the activities. Knowledge of fundamental concepts in computing, robotics or AI is most often required (e.g., [15, 155]), followed by experience with Scratch or other (block-based) programming languages (e.g., [17, 28]), followed in turn by having met the objectives of prior grades or modules (e.g., [75, 137]), and knowledge of mathematics in order to better understand ML and other algorithms (e.g., [104]).

Regarding the types of learning objectives, a distinction can be made between (1) technology understanding and skills, (2) ethical and societal technology implications, (3) design skills, (4) attitudes and mindsets, and (5) other forms of knowledge and skills (see Figure 8). These different types of objectives are combined in varying combinations in the reviewed records. They will be discussed in more detail in the remainder of this section.

Technology understanding and/or skills are mentioned as learning objectives in all but three records (i.e., [95, 97, 120]) (see Table 3). The term "understanding" (or knowledge) refers to familiarity with factual information and theoretical concepts, whereas the term "skills" refers to the



LEARNING OBJECTIVES

Fig. 8. Distribution of included records (n = 107) per technology and type of learning objective.

Table 3. Different Objectives for technology Understanding and Skills Fou	and in the Reviewed Records
---	-----------------------------

Туре	Technology	Learning Objectives	Examples
Understanding	Computing	COMPUTATIONAL THINKING AND CS CONCEPTS/PROCESSES	[50, 70, 82]
Skills	Computing	BLOCK-BASED AND MORE ADVANCED PROGRAMMING	[25, 27, 46]
Understanding	AI/ML	AI AND ML CONCEPTS/PROCESSES AND APPLICATIONS – General understanding of AI and different types of ML – Nuanced understanding of how ML works incl. data analytics and different algorithms	[111, 146] [45, 165]
Skills	AI/ML	BUILDING, TRAINING AND TESTING ML MODELS – GUI; without programming – Incl. block-based programming – Incl. more advanced programming	[18, 66] [73, 166] [88, 113]
Understanding	IoT	IoT CONCEPTS/PROCESSES AND APPLICATIONS	[60, 93, 123]
Skills	IoT	EXPLORING, DESIGNING AND PROTOTYPING IoT – Exploring and designing IoT applications – Prototyping IoT applications (incl. electronics and programming)	
Understanding	AR/VR	AR/VR CONCEPTS/PROCESSES AND APPLICATIONS	[26, 63]
Skills	AR/VR	USING & CREATING AR/VR ENVIRONMENTS – Using AR authoring tools – Creating AR/VR environments (incl. 3D modelling and programming)	[78, 82] [26, 63]

ability to perform a certain task or role and requires the application of knowledge to specific situations. Technology understanding is formulated both in fundamental CS concepts (e.g., [70, 82]) and in terms of the way emerging technologies such as AI/ML (e.g., [111, 165]), AR/VR (e.g., [26]) and IoT (e.g., [93, 123]) work and what their application domains are. Technology skills, in turn, are diverse. They include block-based and more advanced programming (e.g., [25, 27, 46]), building, training and testing ML models (e.g., [88, 166]), prototyping IoT applications, including electronics (e.g., [56, 99]) and creating AR and VR environments (e.g., [26, 78]). Developing these skills often goes hand in hand with deepening the understanding and knowledge of emerging technologies

- and vice versa. Moreover, many researchers do not make a clear distinction between technology "understanding" or "knowledge" on the one hand and the development of practical technical "skills" on the other, approaching these two objectives as one and the same. An example can be found in Williams et al. [158] who introduced the AI concepts of knowledge-based systems, supervised machine learning, and generative AI to young children (ages four to six) by letting them explore and tinker with real examples using the block-based programming platform PopBots.

Remarkably, half of the records that focus on AR and VR (14 out of 27) use the technology as an instructional aid in teaching programming skills, rather than a learning objective in itself. This means that the objective in these records is not to develop students' understanding of how AR and VR work and what their functional strengths and limitations are (e.g., [27, 34, 87]). These records were nevertheless included in the review because their focused objective is technology education, which aligns with the inclusion criteria.

About one-quarter of the records (28 in total) include the ethical and societal implications of emerging technologies as learning objectives. Most of these (24 out of 28) focus on AI/ML. However, this focus on ethical and societal implications is often merely an afterthought, with most attention given to developing students' technology understanding and skills. A few reports refer very broadly to ethical and societal implications without further specification [159]. Others focus on concrete issues such as bias [35], fairness [18], data privacy [129], security [33], accuracy [134], accountability [4], transparency (e.g., [48], and how the technology should or should not be used [85]. Complex issues related to power and democracy are rarely touched upon. One of the few records that foregrounds ethics is Bilstrup et al. [9], who engaged high school students in the design of a supervised ML application that addresses a real-world need in their lives, while simultaneously exploring the ethical implications of the proposed design. During the process, students become aware of how difficult it is to avoid ethical issues, even with the best intentions.

Almost one-quarter of the articles (25 in total) combine technology understanding and/or skills with one or more design skills as learning objectives. "Ideation" is the design skill that is most often mentioned (e.g., [54, 120]), followed by "designing IoT applications" (e.g., [5, 133]), "presenting and providing arguments for design concepts" (e.g., [9, 93]), and "design and creative thinking" (e.g., [17, 123]). "Iterative testing of design concepts" (e.g., [105]) and "game design" (e.g., [82]) are mentioned a few times. Not all records that refer to design skills as learning objectives attribute the same importance to them. Noteworthy is that design skills always co-occur with other learning objectives, especially with technology understanding and/or skills. In many cases, practising design and technical skills go hand in hand, both contributing to students' technology understanding. A good example is provided by Toivonen et al. [134], who developed a two-week program to introduce students to the core theoretical concepts of ML (i.e., training set, prediction accuracy, class label) as well as the practical skills involved in training an ML model to solve a predefined problem. The program includes ideation, user-interface design, iterative testing, and pitching design concepts [134].

Attitudes and mindsets are referred to in almost one-quarter of the records (24 in total) as study objectives in the context of emerging technologies. A recurrent objective in this regard is for students to develop a positive attitude towards and interest in STEM (10 records) (e.g., [57, 141]), AI (e.g., [159]), or digital technology more broadly (e.g., [117]) (8 records). Other, less frequently mentioned attitudes include "active and engaged citizenship" (e.g., [133]), "resourcefulness" (e.g., [148]), and "confidence in one's own abilities" (e.g., [73]). Just as with the presentation of ethical and societal implications and design skills, these attitudes are presented in conjunction with technology-related learning objectives.

This is also the case for other learning objectives (52 records) – for instance, in relation to STEM subjects such as physics and maths (e.g., [51, 119]), or transversal "twenty-first century" skills

applicable in a wide variety of contexts including collaboration (e.g., [23, 104]), critical thinking and reflection (e.g., [22, 142]) and problem-solving (e.g., [5, 151]). Roughly half of the records (52 in total) refer to learning objectives like these in addition to technology understanding and/or skills. A good example is an educational scenario developed by Spyropoulou et al. [129]. This aims at familiarising students aged 14–16 with IoT technologies through a cross-curricular STEM approach including learning objectives related to science (i.e., ultrasonic physics), technology (i.e., Arduino programming and the use of sensors), engineering (i.e., developing and testing IoT applications), and maths (i.e., volume and distance calculation).

In sum, initial steps have been taken towards defining learning objectives with regard to emerging technologies, sometimes in the form of dedicated curricula or cross-curricular approaches. Although learning objectives tend to lack operational detail and multiple progression levels, it is encouraging that technology-specific objectives are often combined with a range of other objectives. This is the case for all three emerging technologies, although clear differences could be discerned in terms of emphasis (see Figure 8 and Table 3). Objectives include the ethical and societal implications of emerging technologies, design thinking, positive changes in student attitudes, a broad range of the twenty-first century or transversal skills such as critical thinking and collaboration, and STEM objectives beyond technology, such as maths and physics. But despite this combination of different types of objectives, a humanities perspective foregrounding the implications and design aspects of emerging technologies is generally either lacking or treated as an afterthought in the learning activities. More holistic approaches need to be developed, which balance technologyspecific objectives with a humanities perspective to promote more comprehensive understandings and skillsets. Related to this, the urgency of "why" students need to attain these learning objectives needs to be explicitly addressed by researchers. In more than half of the records, this is currently not the case. An almost similar proportion of the records present standalone activities that are not, or are insufficiently, integrated in school environments and curricula. These findings are indicative of a rather narrow focus of research on emerging technologies in K-12 education.

4.4 Educational Frameworks and Practices

This section looks into the theoretical learning frameworks and educational practices adopted or proposed for introducing emerging technologies to K–12 students. Educational practices include the format and duration of suggested activities (within a larger setting of formal, non-formal, or informal learning), as well as the pedagogical strategies used to attain the learning objectives explicitly or implicitly referred to in the reviewed records.

The majority of records do not present a solid theoretical learning framework for introducing emerging technologies. Records that integrate learning theory (24 in total, of which 15 relate to AI/ML and nine to IoT) rely on one or more of the following theories: constructionism (e.g., [25, 158]), socio-constructivism [97], actor-network theory [51], participatory and collaborative learning theories [5], situated and experiential learning [93], universal design for learning [111], and design-oriented pedagogy [147].

Of these articles, the majority (13 in total, of which ten relate to AI/ML and three to IoT) rely on constructionism as the underlying theoretical framework, but typically without providing much explanation, using it as a synonym for hands-on activities or learning-by-making (e.g., [73]). In short, constructionism holds that learning happens most effectively when children engage in making tangible objects in the real-world, thereby creating mental models to understand the world around them and using what they already know to acquire more knowledge [107, 108]. Constructionism advocates student-centred discovery learning, whereby students make connections between different ideas and areas of knowledge, facilitated by the teacher through coaching rather than lectures or step-by-step guidance [108].

47:18

47:19

mented constructionism by scaffolding an open-ended creative learning process with a customdeveloped extension for Scratch. Students aged 14 to 17 used the tool called "Let's Chance" to create their own data, models, and possibilities, allowing them to explore powerful ideas of probabilistic thinking and modelling, which in turn helped them to understand how AI technologies make predictions based on training data [35]. Another example is provided by Kandlhofer et al. [75], who state that they developed teaching modules to introduce AI and ML to high school students largely based on principles of constructionism, comprising a wide range of hands-on activities, tools, and platforms as well as different pedagogical strategies including project-based teamwork, storytelling, and peer tutoring [75].

Regarding the format and duration of the suggested learning activities, about one-quarter of all records (24 in total) report on a single workshop or session of one or a few hours (e.g., [43, 44, 167]). A similar number (28 in total) report on a limited number of short sessions, typically three to six sessions of a few hours each (e.g., [78, 92, 118]). Notable exceptions include a small number of records (nine in total) reporting interventions that ran on a regular basis for several months (e.g., [58, 63]) up to a whole year, for instance in the form of a dedicated AI course (e.g., [65, 115, 126]). Three additional records propose AI and ML curricula covering multiple grades, but without having conducted any activities yet [85, 137, 162]. For the remaining records, the format and duration were unclear or not relevant.

Not surprisingly, formal education in primary or high schools is the predominant context for learning activities about emerging technologies (66 records) (e.g., [42, 116, 118]), followed by nonformal education (17 records) such as after-school robotics clubs [29] or workshops [67], science fairs [3], makerspaces [112], and summer schools [155]. Only a few records (five in total) target both formal and non-formal education (e.g., [18, 50]). Two records focus on informal education in a home context, facilitated by parents (e.g., [122, 146]). As discussed in the section on learning objectives, however, the preference for the formal education context does not necessarily mean that the learning activities conceived by researchers are integrated in existing school curricula. Oftentimes, researchers move rapidly in and out of schools to iteratively evaluate learning activities and tools and collect empirical data. A strength, on the other hand, is that most studies are conceived as interventions in the real-world and cover a variety of learning contexts. These findings are consistent across all three emerging technologies under consideration.

Finally in this section, we look into pedagogical strategies used or suggested by researchers. We use the term pedagogical strategies to refer to the ways in which learning content and materials are created and presented to learners – in this case, K–12 students. Detailed information about pedagogical strategies was found in approximately one-fifth of the records (20 in total, of which 12 focus on AI/ML, seven on IoT and one on AR/VR) (e.g., [60, 79, 147]). One-third (34 in total) disclose no information about such strategies, or the code was not applicable; however, some information about pedagogical strategies could be derived indirectly, for instance, by looking at whether learning activities were hands-on or collaborative.

Based on this explicit and implicit information in the reviewed records, 15 distinct pedagogical strategies could be discerned (see Table 4), of which three are predominant: active and engaged teaching, as opposed to passive listening (e.g., [52, 100, 125]), small group work and peer learning (e.g., [1, 68, 88]), and technology-mediated teaching by letting students use, modify, and/or construct technology artefacts. This last strategy includes block-based and more advanced programming activities (e.g., [34, 87]), developing IoT applications (e.g., [56, 99]), building or modifying and testing ML models (e.g., [88, 166]), and creating AR games [78] and VR environments [26].

Among other pedagogical strategies used in varying combinations are: low-entry barriers to student participation in activities (e.g., [48, 117]), inquiry- and project-based approaches in which

Pedagogical strategy	Description	Examples
Active and engaged learning	Students are active and engage in hands-on activities, as opposed to passive listening	[52, 100, 125]
Technology-mediated	Learning is facilitated through the use, modification, and/or construction of technology artefacts	[87, 99, 166]
Collaboration and peer learning	Students collaborate, often in small teams, and facilitate each other's learning or act as tutors for peers	[1, 68, 88]
Low-entry barriers	Students with little prior knowledge or experience are enabled to participate in learning activities	[36, 48, 117]
Inquiry or project-based	Projects or inquiries with multiple pathways towards a solution drive students' engagement and learning	[56, 60, 129]
Design-process driven	esign-process driven Students are guided through a design cycle incl. field study, problem (re)framing, ideation, prototyping, testing, and so on.	
Tinkering and creative exploration	ve Free exploration and tinkering with creative materials and learning contents	
Reflective practice	Students discuss and critically reflect on their learning trajectory, practices, and/or technology artefacts	[9, 22, 120]
Authenticity and closeness	Real-world and personally meaningful problems are used as a starting point for learning activities	[79, 133, 147]
Learner-centred or self-guided	Students direct their own learning and can pursue their own interests in relation to the topic	[26, 58, 71]
Knowledge-driven	New topics or concepts are introduced through (short) lectures, often complemented with hands-on activities later	[104, 119, 149]
Unplugged approach	*	
Embodied learning		
Crossdisciplinary	Students are introduced to topics from different disciplinary perspectives and across school subjects	
Modular and adaptive	Learning content is broken into different parts that can be taught independently and adapted to students' current level	[59, 65, 75]

Table 4. Frequently used Pedagogical Strategies Embedded in Learning Activities about Emerging Technologies Across Age Cohorts in K-12

students engage with open or wicked problems (e.g., [56, 60]), design-process driven activities (e.g., [109, 133]), creative exploration and tinkering with learning materials and content (e.g., [35, 158]), reflective practices (e.g., [9, 120]), an emphasis on authenticity and closeness by structuring activities around real-world and personally meaningful challenges (e.g., [79, 133]), self-guided or student-centred learning (e.g., [26, 71], knowledge-driven approaches introducing new concepts with (short) lectures, often complemented with hands-on activities later to apply this new knowledge (e.g., [119, 149]), unplugged activities in which the use of digital technology is deliberately avoided (e.g., [48, 81, 103]), embodied learning or using one's body via actions and gestures to create new knowledge (e.g., [126, 167]), crossdisciplinary perspectives that are not confined to traditional subject boundaries (e.g., [43, 142]), and modular and adaptive activities (e.g., [59, 65]) (see Table 4 for an overview). Although these pedagogical strategies are used across the three types of emerging technologies, no distinct patterns could be discerned. The reviewed records also provide little information about the suitability of these strategies for particular age cohorts, although generally speaking less emphasis is placed on cognition (see strategies "knowledge driven" and "reflective practice") and self-efficacy (see "self-guided" and "inquiry or project based" learning) when younger pupils are targeted in lower primary school.

An excellent example in which multiple pedagogical strategies are integrated is provided by Byrne et al. [17]. The authors frame their research on teaching IoT to high school students within a social constructivist framework: more specifically, the Bridge 21 model for collaborative, studentcentred, technology-mediated, hands-on, and project-based learning, which aims at supporting an innovative, twenty-first-century learning environment within schools. With these pedagogical strategies in mind, they developed a four-day hackathon event in which students expand their domain and technical knowledge, investigate a design challenge and IoT possibilities, come up with ideas, and develop a working prototype, realise a digital media campaign to promote their concept, and finally, pitch and critically reflect on their work. In this approach, the development of technical and twenty-first-century skills such as problem solving, communication, and teamwork are integrated [17]. Rattadilok et al. [111], in turn, proposed an active and engaged approach to introduce basic ML concepts to students with little interest in technology. To motivate students, they used the pedagogical strategy of "closeness" by using a well-known and popular mobile game, Clash of Clans, among students as an object of study and experimentation. Students collaborated in small teams to develop game strategies on a worksheet, and fed these to a game bot called "iGaME" (In Class Gamified ML Environment) that used the input a predefined number of times, after which it created an output file about its success rate. Students iteratively improved their strategies and competed with other teams to find who had developed the most successful strategy. The session concludes with a class discussion on lessons learned [111].

In sum, the format and duration of learning activities typically ranges from a single workshop to a few sessions of one or more hours. The preferred setting for these activities is formal education, followed by non-formal education, including after-school robotics clubs, science fairs, maker spaces, and summer schools. To scaffold students' learning in these settings, a wide range of pedagogical strategies are suggested in the literature, among which three are predominant: active and engaged learning, collaboration, and technology-mediated teaching by letting students use, modify, or construct technology artefacts. These are complemented with a range of other pedagogical strategies in varying combinations such as learner-centred and inquiry-based teaching, authenticity and closeness, tinkering, and reflective practice. These findings are more or less consistent across all three types of emerging technologies. Remarkably, only one-fifth of the records – none of which focus on AR/VR – connect these strategies to the existing corpus of learning theory, and often in an ad hoc manner. This lack of clear and well-developed theoretical trajectories hampers the advancement of this new research field.

4.5 Technology (and Other) Tools for Learning

This section looks into technology and other tools developed or used by researchers to support learning activities about emerging technologies. Here we distinguish between unplugged tools, tools that incorporate emerging technologies, and any other digital technology tools. The section first looks into tools used to teach or introduce AI and ML, followed by IoT, and finally AR and VR.

Only 16 records do not present or refer to new or existing technology tools – no surprise, as the topic of interest is technology education. From the 57 reports that focus on AI or ML education, 42 present technology tools, often in the form of ML-powered tools specifically designed to teach ML and referred to in this section as "ML tools." What these ML tools have in common is that they allow students to develop, train, and test models, be it with different types of input data including, among others, images (e.g., [88, 122, 155]), sounds (e.g., [86]), and gestures (e.g., [67]). A good example of a custom-developed ML tool is the iOS application AlpacaML, which facilitates the construction and use of ML gesture models based on data from wearable sensors [166, 167]. With this tool, students create, improve, and test models of their own sports-related gestures and get

real-time feedback (e.g., about a bad or good soccer pass). By using students' sport expertise and identity as a point of departure, the authors hope to foster curiosity about ML [166].

Another characteristic of ML tools is that they expose or glass-box some ML concepts or processes while black-boxing others, often to lower the barrier for novice learners. This is, for instance, the case with the gesture-based supervised ML tool developed by Hitron et al. [66]. The ML tool aims at familiarising students with data-labelling aspects (i.e., sample size, sample versatility, and negative examples) and model evaluation, while deliberately black-boxing other aspects such as feature extraction, model selection, and validation to reduce students' cognitive load [66].

ML tools differ in their degree of complexity in operation and, related to this, the possibilities they offer to explore and learn about ML. A distinction can be made between ML tools using **graphical user interfaces** (**GUIs**), low-code, or block-based coding environments, and more advanced programming environments. GUIs provide the lowest entry level as they require no coding to train, test, and execute predefined ML models (e.g., [111, 117, 147]). An example of an easy-touse GUI is **Google Teachable Machine** (**GTM**). Vartiainen et al. [146] used GTM to introduce young children (ages 3–6) with no programming experience to ML. GTM is a web-based ML system powered by classification algorithms such as convolutional neural networks. It allows people to quickly train their own ML models, without programming, using images, gestures, and sounds as predictive modes. Children used GTM to create models for three different facial expressions, then explored the relationship between input (facial expressions) and output (sounds and images) [146].

ML tools that adopt block-based programming environments are more challenging to operate, but offer more possibilities in return. These ML tools utilise a visual drag-and-drop learning environment whereby students use coding instruction "blocks" to develop simple ML applications. Most often used are extensions for Scratch (e.g., [1, 35, 113, 157]), Snap! [73], and App Inventor [112]. García et al. [52], for instance, developed an educational resource to teach ML in schools with ML4K (Machine Learning for Kids), a web platform for children to build ML models that can be exported to Scratch or App Inventor to develop ML-powered applications [52].

In a few records, students use more advanced programming environments such as Python, C++, and/or Java to develop ML-powered applications (e.g., [43, 88]) or autonomous robots and vehicles (e.g., [58, 68]). Since these tools are technically complex, prior knowledge of computing is usually required. To extend the possibilities, block-based and more advanced programming environments are sometimes combined with general ML platforms and open-source libraries such as ExpliClas [4], WatsonAI [148], Tensorflow [100], and AzureML Studio [159].

In addition to ML tools, a range of complementary digital tools are used to support K–12 students' learning about AI and ML, including Lego WeDo 2.0 [158], Microsoft Excel [46], and video communication and cloud computing tools [68]. A few reports (six in total) deliberately use unplugged tools to provide alternative pathways to develop an interest in and understanding of AI and ML. These studies often target students with little motivation for or expertise in technologyrelated subjects. Examples include a card-based design game to introduce AI ethics [9], a manmachine simulation game [103], and a Turing roleplaying activity [81].

Of the 28 records that focus on IoT education, 27 present or refer to the use of technology tools. The characteristic of these tools is that they combine existing with custom-made components and modules, often in the form of open-source IoT toolkits. IoT toolkits usually include a range of tangible components, especially compared to ML tools that are often primarily web-based. Typical components are microcontrollers, programmable sensors and actuators, software, and coding environments including brands such as Arduino, Udoo, Raspberry Pi, ThingsBoard, Micro: bit, CloudBits, Bee-Bots, Lego Mindstorms NXT, and Cubelets (e.g., [5, 25, 33, 56, 99]). Important to

note is that some authors use the term IoT interchangeably with 'physical computing,' thereby disregarding the cloud component (e.g., [54]).

Educational Lab Kit is an example of such an IoT toolkit [98]. It is an open-source platform that combines widely available and custom-made hardware electronics that are Arduino- or Raspberry Pi-compatible with web-based tools, gamification elements, and activity guides. With the toolkit's sensing devices and cloud infrastructure, students can collect real-time data from their school buildings and use it in maker-like STEM activities in a context of energy awareness and sustainability [98]. Another example is the UMI-Sci-Ed toolkit that consists of an Udoo-Edu hardware kit with an accompanying online programming environment and different educational scenarios [57]. The Udoo-Edu hardware kit is packed in a suitcase and includes an Udoo Neo Board with USB, ethernet, and Wi-Fi, a 1 GB RAM processor, and multiple sensors and actuators. The educational scenarios provide teachers with all the necessary components to facilitate students in exploring and developing IoT applications, including design challenges (e.g., smart recycling), learning objectives, activities and materials [60].

Overall, IoT toolkits require prior knowledge of electronics and programming to operate, both for students and facilitators. Substantial efforts have nevertheless been made to provide low barriers to entry for novice learners, not least through the development of easy-to-use GUIs and software platforms to program IoT components (e.g., [77, 150]). Setiawan et al. [124], for instance, developed a visual mobile programming tool for IoT applications with Raspberry Pi, which students with no programming knowledge or skills should be able to use. Along the same lines, Vakaloudis et al. [141] introduced a new software platform that enables teachers with no experience to integrate IoT in STEM education, and Rizzo et al. [112] developed an extension for the App Inventor (UAPPI) that transforms it into a GUI to program physical objects. Arora et al. [5], in turn, developed a physical and digital toolkit called DIO that consists of custom-made dome-shaped modules with an embedded assortment of input and output functionalities. The modules can be easily programmed through an AR-based application leveraging 3D identification patterns present on each module. This way, DIO facilitates children to develop multiuser wearables and environmental interaction designs [5].

Only one record relies exclusively on unplugged learning tools for IoT education. The Tiles IoT Inventor Toolkit, developed by Mavroudi et al. [93], enables children to generate ideas for IoT ecologies within a specific domain and without the use of technology. In other records, IoT toolkits are sometimes combined with unplugged tools, such as educational scenarios and/or activity guides (e.g., [56, 98, 129]), but these tools merely have a supporting role.

All 27 records focusing on AR and VR present technology tools. Five of these are VR tools, 22 AR tools. AR and VR tools are typically custom-made with technologies such as Unity 3D, Vuforia Engine, TopCode, EasyAR, and Google ARcore. They can be grouped into tools that aim at developing students' computational thinking and programming skills (18 records), or alternatively, authoring tools to create AR and VR (game) environments (seven records).

Characteristic of the largest group, here referred to as AR and VR programming tools, is that these tools do not scaffold learning about AR and VR, but rather, use AR and VR as instructional aids to provide real-time visual feedback during programming and related tasks (e.g., [34, 44]). A good example is the mobile visual programming environment for the Thymio II robot, which runs on Android and iOS [87]. Students use the environment to solve increasingly challenging tasks, while learning robot programming and event handling. Another example is the AR game CodeCubes, which combines physical programming with simultaneous visualisation to promote an interactive and engaging learning experience [28, 29]). Similar approaches are used for tools such as AR maze [70], CodAR [125], CodeBits [61], HyperCubes [50], and ARQuest [53].

The second group, authoring tools, enable students to build their own AR games (e.g., [109]) or VR environments (e.g., [63]). Examples of AR authoring tools include TaleBlazer, an AR platform to make geolocation games [109], and the interactive storytelling platform ARIS, which consists of a web-based editor and a client-based app to develop AR games with physical objects in a specific location [82]. Examples of VR authoring tools include [63], who developed a low-cost VR-driving simulator and graphical programming interface, among others (e.g., [26, 91, 92, 95]).

In sum, a wealth of digital learning tools to introduce emerging technologies to K-12 students have been developed. ML tools are typically designed to glass-box some aspects of the technology while black-boxing others, and they differ in degree of complexity and possibilities. GUIs and block-based coding environments are most often used to enable students to develop, train, and test simple ML models, but in a few cases, students engage in more advanced programming to develop ML-powered applications. IoT toolkits are often open-source and combine existing with custom-made components and modules to enable students to design and develop IoT solutions. As with ML tools, IoT toolkits do not glass-box all processes in order to manage complexity. Although IoT toolkits require some prior knowledge in programming and electronics, easy-to-use GUIs and software platforms have been developed over the years to make programming IoT components more accessible. AR and VR tools, finally, are predominantly programming tools that do not necessarily enable students to learn about AR and VR, with an additional small group of authoring tools to create AR and VR (game) environments. The development of learning tools for AR and VR, their characteristics, and their implications provide opportunities for future research. Another interesting line of research, found in only a few records, is the use of unplugged tools to engage students with little interest and few prior skills in digital technology in learning activities.

4.6 Empirical Evaluation and Student Assessment

This last section presenting our results looks into the ways in which technology tools and teaching approaches are empirically evaluated, whether and how students' learning is assessed either formatively and/or summatively, and the degree of constructive alignment between learning objectives, activities, and student assessment. This section furthermore provides insight into the methods used to collect empirical evidence, and it provides examples of typical findings reported in the included records.

Of the 107 reviewed records, 65 present original empirical data, of which 30 focus on AI/ML, 18 on IoT and 17 on AR/VR. The majority of these include studies evaluating learning activities (55 records) and tools (50 records) (see Figure 9). Activities and tools are often evaluated in tandem (38 records), and these constitute the main focus of the record (e.g., [52, 96, 100]). This is especially the case with records that introduce tools for learning about emerging technologies. Examples include Glaroudis et al. [57], who evaluated a new open-source learning environment deployed in an inquiry-based and collaborative learning approach to introduce IoT, as well as Lindner et al. [81], who evaluated whether unplugged tools and hands-on activities are suited for teaching AI in a comprehensive way to high school students.

From an in-depth inspection of the quality criteria used to evaluate technology tools, three main criteria surface: usability (e.g., [25, 34, 124]), students' perception of tools (e.g., [29, 42]), and the ways in which they interact with said tools (e.g., [50, 146]). However, less studied are the particular aspects and mechanisms of technology tools (e.g., balancing the glass- and black-boxing of features) and how they contribute to students' learning.

For learning activities, the quality criteria are more diverse. They include students' engagement (e.g., [54, 74]) and collaboration with team members (e.g., [23, 53]), students' learning experiences, for instance in relation to factors such as "satisfaction" and "enjoyment" (e.g., [56, 166]) or with regard to how they perceive the activities and content (e.g., [15, 88]), and finally, possible shifts



Fig. 9. Distribution of records presenting original empirical data (n = 65) per technology, based on empirical evaluation of learning tools, approach, and/or students' learning.

in students' interests and attitudes that can be attributed to the activities (e.g., [60, 126]). These quality criteria for evaluating learning activities are sometimes combined in a single study, as in Schneider et al. [123], who used observation notes to study students' engagement during the activities and a post-questionnaire to collect information about their experiences. This resulted in nuanced findings about students' learning process, the perceived workload, and how the activities unfolded [123].

This example furthermore shows how different qualitative data collection methods are combined to evaluate learning activities and tools. The most frequently used techniques are open and closed questionnaires (e.g., [117, 118]), semi-structured exit interviews (e.g., [23, 28]), and participant observation (e.g., [53, 115]). Mixed-methods studies, in which qualitative and quantitative data are triangulated, are rare (e.g., [129]).

From the 65 records presenting original empirical data, slightly more than half (36 records) include some form of assessment of students' learning (e.g., [15, 67, 105]) (see Figure 9). Assessment of students' learning is never the sole focus, and always takes place in conjunction with an evaluation of learning activities and/or tools. Furthermore, assessment is typically summative in nature, which means that students' learning is evaluated at the end of an instructional unit by comparing it against a standard or benchmark (e.g., with a pre- and post-test). Summative assessment serves the purpose of accountability, ranking or certifying competence [10]. This contrasts with formative assessment, which summarises students' development at a particular point, and primarily aims at promoting students' learning [10]. Surprisingly, though, no records could be identified that focus on formative assessment or, related to that, feedback and feed-forward practices to improve students' learning. This does not mean that such practices did not occur during the learning activities, either spontaneously or deliberately, but a formative assessment was not an explicit research objective in these studies.

In the reviewed records, summative assessment is directed towards students' knowledge and understanding of technology concepts and processes (e.g., [5, 67, 74]), and to a lesser extent, students' skill development, including technical skills (e.g., [34, 82]), design skills (e.g., [54, 134]),

and problem-solving skills (e.g., [75, 93]). Pre- and post-tests are most often used to assess students' knowledge acquisition (e.g., [67, 105, 118])), whereas artefacts (i.e., worksheets, models, prototypes) and students' complementary explanations of these artefacts are used to gain insight into students' skill development and the ways in which they applied and further developed new knowledge by engaging with practical problems (e.g., [5, 117]). Students' self-perceived learning, in turn, is typically captured with open-ended questionnaires and semi-structured exit interviews (e.g., [15, 98]).

A related finding is that approximately half of the 36 records that include student assessment showcase constructive alignment. Constructive alignment is a principle devised by Biggs [7, 8] which refers to developing learning activities and assessment tasks that directly address the intended learning objectives. In the reviewed records, alignment often breaks down because assessment criteria do not or only partly align with pre-set learning objectives (e.g., [46, 82, 105]). Even with constructive alignment established, learning objectives may be described in high-level or vague terms, making it hard to judge whether these objectives are indeed aligned with the proposed learning activities and assessment tasks (e.g., [54, 91, 167]). There are, however, a few good examples of how constructive alignment can be established when introducing emerging technologies to K–12 students (e.g., [117, 149]).

One such example is provided by Hitron et al. [66] who showcase how they assessed students' understanding of three data-labelling aspects (i.e., sample size, negative examples, and versatility) in relation to classification problems in supervised ML. These learning objectives were set in advance and taught in three different activities, one for each main objective. The assessment procedure consisted of a pre-test before the learning activities, a short interview with students in which they explained the process they went through, a post-test in which they were given two ML examples (one similar to the learning experience and one different) and had to explain the underlying data-labelling processes, and two open follow-up questions to gauge whether students could relate the learning content to their own lives. This straightforward, yet effective procedure shows constructive alignment in practice.

Although providing a qualitative meta-analysis of empirical findings is not the focus of this review, it is apparent that most of the records present positive findings. Only seven records were identified as providing greater nuance by including unexpected or negative findings, for example in terms of students' learning, unforeseen obstacles (e.g., costs, public support, teacher motivation), or activities and tools that did not lead to desired outcomes (e.g., [42, 56, 123]). As for the majority of records, it is unclear whether there were indeed no unexpected or negative findings to report, or whether such findings were simply disregarded. Typical examples of positive findings include increased motivation among students to learn abstract concepts [92], design guidelines to build tools for learning about emerging technologies [82], positive attitudes towards (careers in) STEM [129], students being capable to build, train, and evaluate simple supervised ML models [167], good usability and acceptance of new technology tools [119], high effectiveness of a proposed curriculum and teaching platform [158], and so on.

In sum, of the 65 records that present original empirical data, the majority evaluate learning tools and activities, often together. An obvious strength of current research is the use of a range of qualitative methods to study different aspects of students' learning experiences in real-world settings. However, the results mainly provide a good news show with only a few records reporting unexpected or negative outcomes – a feature indicative of the lack of maturity of this nascent field. Quality criteria to evaluate tools are directed towards usability and students' perception of and interaction with these tools, but less towards the underlying mechanisms of these tools and how these contribute to students' learning. For learning activities, quality criteria include students' engagement and collaboration, their learning experiences, and possible shifts in attitudes. A

remarkable finding is that only half of these 65 records include summative assessment of students' learning. Summative assessment is narrowly directed towards students' technology skills and understanding, leaving out the ethical and societal implications of emerging technologies as well as other aspects. Other remarkable findings are the lack of constructive alignment between learning objectives, activities, and assessment tasks, and the fact that no research on formative assessment could be identified in the literature.

To recap, the results of this systematic mapping review were structured along the five central topics of interest, preceded by a description of the dataset. The findings show a sharp increase in interest in teaching emerging technologies in K–12 education, especially in recent years, but many challenges remain unaddressed. The next section discusses our main findings, leading to our presentation of a future HCI research agenda to advance and mature the field.

5 DISCUSSION

In the past decade, research addressing the need to teach children about digital technology, digitalisation and, more recently, emerging technologies has increased on a global scale (e.g., [102, 129, 136, 137]). But teaching about emerging technologies has only now begun to appear beyond academia as a topic for education, and it makes up a very small share compared to digital technology and digitisation. Although the emerging technologies under consideration in this article (AI, ML, IoT, AR and VR) have reached a certain degree of coherence and momentum, as witnessed in increased technology maturity and growing numbers of applications, their full impact on society and on children's future lives is still uncertain and ambiguous [64, 114]. This poses unique challenges for K–12 education, as these technologies are hard to comprehend due to their complex and distributed nature, which allows little transparency into their functions and implications. It is, however, critical that children move beyond passive consumption, and acquire the competences to maximise the opportunities associated with emerging technologies while limiting exposure to harm and risk [19, 24, 84]. This may, in turn, contribute to children's computational empowerment [39] and the development of an *emerging technology literacy* – which will give them a say in a society with increased human-computer interaction [144, 145].

Research on emerging technologies in K–12 education stems from a diverse range of research fields – from novel technologies (human-computer interaction) to technology education and learning (computing education and learning sciences) and design for children (child–computer interaction). This systematic literature review has provided an overview of the state of the art across these fields, including 107 records published between 2010 and 2020. The literature was analysed covering central topics of (1) target groups and teacher roles, (2) learning objectives and curricular implementation, (3) educational frameworks and practices, (4) technology and other learning tools, and (5) empirical evaluation and assessment of students' learning. The findings show the urgent need on the global scale for inter- and transdisciplinary work that integrates such aspects into a more coherent and developed field of research and practice. In this section, therefore, we first provide a concise summary of the most important findings per central topic, followed by nine propositions for developing a shared agenda for future research (see Section 5.2 and a summary in Table 5).

5.1 State of the art of Emerging Technologies in K-12 Education

The findings of this systematic mapping show a rapid increase in interest in emerging technologies in K–12 education on a global scale in the past few years, across a broad range of venues, especially for AI and ML (see Section 4.1 – Figure 4). However, the vast majority of the 107 records retrieved from Scopus and through citation searching originate from Europe and North America, showing a limited and unequal geographical distribution (see Figure 5).

No	Recommended research agenda	Cross-reference results
1.	Explicitly state the importance of teaching about emerging technology in K–12 and provide detailed and progressive learning objectives	Section 4.3. Learning objectives
2.	Foreground ethical and societal implications to offer a more comprehensive understanding of emerging technologies	Section 4.3. Learning objectives (Figure 7)
3.	Identify underlying mechanisms of both digital and unplugged learning tools, and develop tools to learn about and design with AR/VR	Section 4.5. Technology and other learning tools
4.	Target a more diverse range of K–12 students, leading to more inclusive approaches to introduce emerging technologies	Section 4.2. Target group (Figure 5)
5.	Actively engage teachers in both backstage work and facilitation, and coordinate professional development programmes	Section 4.2. Teacher roles
6.	Integrate learning activities in formal school environments and develop cross-curricular approaches beyond STEM subjects	Section 4.3. Curricular implementation (Figure 6)
7.	Provide an explicit focus on pedagogical strategies grounded in existing (and new) learning theories	Section 4.4. Educational frameworks and practices (Table 4)
8.	Consider formative assessment strategies and how these can inform feedback and feed-forward practices	Section 4.6. Student assessment
9.	Conduct long-term empirical studies to generate nuanced findings about emerging technologies in K–12 education	Section 4.6. Empirical evaluation (Figure 8)

 Table 5. Overview of the Nine Trajectories for a Future HCI Research Agenda, Cross-referenced with the Results Sections

Learning objectives (see Section 4.3) in the reviewed records prioritise technology skills and the understanding of core concepts and processes (see Table 3). Most of the studies require little or no prior experience from students. Technology-related objectives are usually combined with other STEM objectives, while a humanities perspective incorporating design aspects and ethical and societal implications of emerging technologies is largely missing (see Figure 8). The same goes for the higher-order objectives that motivate why children should develop technology and other competencies (e.g., preparation for STEM careers, a broad literacy perspective). In more than half of the records, these could not be identified. Nearly half of the records present standalone learning activities, without integration in or across existing (and new) curricula (see Figure 7). These two characteristics – the high ratio of standalone activities prioritising technology-specific learning objectives, and the lack of curricular integration and different learning progression levels – limit the depth and impact of the research.

When looking at practices to scaffold children's learning towards pre-set objectives (see Section 4.4), it is apparent that few studies ground learning activities in established learning theories, or that they do so in a rather superficial manner, especially in relation to constructionism. The format and duration of activities typically ranges from a single workshop to a few sessions in formal or non-formal learning settings, with grades 8 to 10 as the most common target group (see Section 4.2 – Figure 6). Further, even in the studies conducted in a school context, teachers are given a marginal role to play in developing, facilitating, and evaluating learning activities, with researchers retaining a firm grip on what should be taught and how. Teachers' limited involvement, and the fact that most interventions are short-term and include a rather narrow range of children, offers potential for future research.

A variety of pedagogical strategies have been used to engage and motivate students in learning activities. Hands-on, collaborative, and technology-mediated teaching are frequently used strategies (see Section 4.4 – Table 4). The characteristic of technology tools to support learning (see Section 4.5) is the way in which they are designed to glass-box some aspects of emerging technologies while deliberately black-boxing other aspects to reduce complexity. Another characteristic is the diversity of easy-to-use GUIs and block-based coding environments: for instance, for

developing, training, and testing supervised ML models or programming IoT components. These tools, combined with a wide range of pedagogical strategies, have successfully lowered the barrier for novice learners to engage with emerging technologies. However, despite the available variety of pedagogical strategies and digital learning tools, the literature shows limited integration in the existing corpus of learning theory.

Of the 107 included records, 65 present original empirical data (see Section 4.6). This is typically qualitative in nature, stemming from 34 students on average per study. The majority of these studies focus on evaluating learning tools and activities based on diverse quality criteria, usually a combination of user experience and learner engagement (see Figure 9). Students' learning is assessed, in a summative way, in fewer than half of these 65 records. Only a few studies show clear constructive alignment between learning objectives, activities, and assessment tasks. This one-sided focus on evaluating learning tools and activities, often at the expense of assessing students' learning, indicates a lack of maturity of this nascent field of research.

5.2 Nine Trajectories for a Future HCI Research Agenda

Based on our extensive mapping of the literature, it is evident that education about emerging technologies in K-12 is not yet an established research agenda. Rather, the mapping reveals dispersed discussions across different research fields. Our literature review points to a range of potentials and discussion points that need to be addressed in order for this important research field to consolidate and mature. The HCI community is ideally positioned to take a leading role in this endeavour and to act as a mediator between neighbouring fields due to its expertise in human-centred approaches to technology and aspects of learning (both ed-tech and tech-ed). Based on the findings, we propose the following global research agenda (see Table 5 for an overview):

- (1) Explicitly state the importance of teaching about emerging technologies and their characteristics and implications in K-12 education. This would clarify researchers' objectives, whether in preparing students for a career in STEM or in achieving the broad literacy considered to be important for all students, whatever their career prospects may be. Connecting this high-level objective or big 'why' question to concrete learning objectives with multiple progression levels would furthermore address the lack of a developmental perspective to learning found in the literature.
- (2) Foreground the ethical and societal implications of emerging technologies, including complex issues related to power and democracy, and including the fact that the design of technology is never value-neutral. This would equip students with a more comprehensive and holistic understanding of emerging technologies, inclusive of how new and unanticipated ethical and societal concerns increase over time, while the uncertainty with regards to possible uses, outcomes, and meanings associated with an emerging technology decrease.
- (3) Identify which underlying mechanisms of digital tools contribute to students' learning about emerging technologies, generating intermediate-level knowledge that transcends the design and evaluation of a particular tool. This would lead to a more nuanced understandings of digital tools and how they can be successfully integrated in learning activities and curricula, addressing a current gap in research. Other research opportunities are the development of learning tools for AR/VR and unplugged tools that lower the barrier for students with little interest in understanding technologies.
- (4) Include a broader range of students beyond lower secondary education (i.e., grades 8 to 10) towards all levels of K-12 education, and with a special focus on underrepresented target groups such as girls, students of diverse socioeconomic and cultural backgrounds, and special education students. This would lead to a more inclusive approach, allowing for

diversity across age, gender, and interests to be represented in the literature. Introducing emerging technologies from an early age could furthermore promote a developmental perspective to learning.

- (5) Actively engage teachers in the development, planning, and facilitation of learning activities so as to install a process of mutual learning between researchers and teachers in line with human-centred and participatory design approaches. Related to this, set up professional development programmes for teachers to scale-up and sustain research-led initiatives. Although initial steps have been taken in this regard, considerable effort is needed to increase the overall impact of research on emerging technologies in K-12 education.
- (6) Carefully integrate learning activities and tools in school environments, as well as in existing and new curricula. Moreover, let K-12 students engage with different emerging technologies across subjects, also outside traditional STEM subjects such as maths and physics. This would not only provide a richer learning context for different types of students, but also avoid the frequently encountered pitfall of standalone learning activities that are insufficiently adapted to the specific characteristics of formal and non-formal learning environments.
- (7) Include the development of different pedagogical strategies in research efforts (see Table 4 for an overview), ground them in the existing corpus of learning theory and, if possible, contribute to the development of (new) learning frameworks in relation to emerging technologies. A wide variety of pedagogical strategies are currently used in changing combinations, often loosely connected to constructionism, but they fail to provide a deeper understanding of how these particular strategies relate to theory and scaffold students' learning.
- (8) Consider not only summative but also formative assessment strategies that prioritise students' individual progress as an integral part of learning activities about emerging technologies. The current literature focuses predominantly on summative assessment, evaluating students' learning at the end of an instructional unit rather than promoting learning through feedback and feed-forward practices based on continuous formative assessment. Providing teachers with formative assessment strategies could furthermore accelerate the integration of emerging technologies in K-12 education.
- (9) Conduct long-term studies and provide nuanced empirical findings about the deployment of learning activities and tools in different educational settings and the ways in which these scaffold students' learning about emerging technologies. Most of the current research consists of short-term qualitative studies, often with a one-sided focus on user experience and student engagement. Moreover, studies that focus on constructive alignment of learning objectives, practices, and tools to scaffold K-12 students' learning are rare.

Bødker and Kyng [12] argued that the HCI community needs to take responsibility in the large contemporary issues and pursue social and political agendas beyond the technological. They point specifically to the need for research on children's engagement with technology, arguing that developing children's future skills in the digital domain is a useful way of opening up democratic debates precisely because technology is often considered an expert domain for the few, with children as easy targets for manipulation [12]. The proposed trajectories for a future research agenda are a response to this and similar calls. This research agenda, if realised over time on a global scale, would build the foundations for a mature inter- and transdisciplinary research field of emerging technologies in K–12 education. We encourage the HCI community to act as a catalyst in the realisation of this agenda. One way forward is to expand its methodological toolbox and initiate

action-research-like programmes in collaboration with learning scientists, media studies scholars, child-computer interaction researchers, as well as other stakeholders. In this way, the HCI community would create a profound impact on children's agency in a digitalised society that reaches beyond academia into policy and education.

6 CONCLUSION

There is an urgent need on a global scale for research on approaches that address how to teach children about digital technology, digitisation, and, more recently, emerging technologies such as AI, ML, IoT, AR, and VR, to which children are increasingly exposed. Calls for action have been raised across policy, within STEM and informatics education, and in academia more generally to integrate and advance the benefits of computational thinking, digital literacy, and maker education for the coming generations.

The findings of the systematic mapping review presented in this article show a sharp increase in interest in education about emerging technologies in K-12, especially in recent years. Meaningful research has been conducted, generating a wide variety of digital learning tools, activities, and pedagogical strategies which have been empirically evaluated in both formal and non-formal learning environments. Initial steps have also been taken towards developing dedicated curricula and cross-curricular approaches that engage proactively with emerging technologies and their characteristics across school subjects. However, many challenges remain unaddressed. Among other things, there is a need for more inclusive approaches and active involvement of teachers, detailed learning objectives distinguishing between different progression levels in connection to school curricula, contextualised approaches that foreground ethical and societal implications of emerging technologies, better integration of pedagogical strategies with the existing corpus of learning theory, and long-term studies that move beyond a focus on technology tools. Based on our findings - and on the gaps in the current literature - we have proposed nine trajectories for a future research agenda (see Table 5). Inter- and transdisciplinary work will be required to realise this agenda on a global scale and to advance and mature this nascent research field. We have argued that the HCI community should take a leading role and act as a mediator with neighbouring fields in the realisation of this agenda. This would create a profound and lasting impact on children's agency in a society with increasing human-computer interaction.

REFERENCES

- Adam Agassi, Hadas Erel, Iddo Yehoshua Wald, and Oren Zuckerman. 2019. Scratch nodes ML: A playful system for children to create gesture recognition classifiers. In *Proceedings of the Extended Abstracts of the 2019 CHI Conference* on Human Factors in Computing Systems. Association for Computing Machinery, New York, NY, 1–6. DOI: https:// doi.org/10.1145/3290607.3312894
- [2] Mariana Aki Tamashiro, Maarten Van Mechelen, Marie-Monique Schaper, and Ole Sejer Iversen. 2021. Introducing teenagers to machine learning through design fiction: An exploratory case study. In *Proceedings of the Interaction Design and Children*. Association for Computing Machinery, New York, NY, 471–475. DOI: https://doi.org/10.1145/ 3459990.3465193
- [3] Jose M. Alonso. 2019. Explainable Artificial Intelligence for Kids. Atlantis Press, 134–141. DOI: https://doi.org/10.2991/ eusflat-19.2019.21
- [4] Jose M. Alonso. 2020. Teaching explainable artificial intelligence to high school students. International Journal of Computational Intelligence Systems 13, 1 (2020), 974–987. DOI: https://doi.org/10.2991/ijcis.d.200715.003
- [5] Jatin Arora, Kartik Mathur, Manvi Goel, Piyush Kumar, Abhijeet Mishra, and Aman Parnami. 2019. Design and evaluation of DIO construction toolkit for Co-making shared constructions. *Proceedings of the ACM on Interactive, Mobile, Wearable and Ubiquitous Technologies.* 3, 4 (2019), 127:1–127:25. DOI: https://doi.org/10.1145/3369833
- [6] S. Ayllón, M. Barbovschi, G. Casamassima, K. Drossel, B. Eickelmann, C. Ghetau, T. P. Haragus, H. B. Homarsdottir, C. Hyggen, O. Kapella, A. Karatzogianni, S. Lado, D. Levine, T. Lorenz, L. Mifsud, D. Parsanoglou, S. Port, M. Sisask, M. Symeonaki, and G. Teidla-Kunitsön. 2020. *ICT usage Across Europe. A Literature Review and an Overview of Existing Data*. Universitat de Girona. Retrieved February 26, 2022 from DOI: 10.6084/m9.figshare.12906737

- John Biggs. 1996. Enhancing teaching through constructive alignment. *Higher Education* 32, 3 (1996), 347–364.
 DOI: https://doi.org/10.1007/BF00138871
- [8] John Biggs. 2011. Teaching For Quality Learning At University (4th. ed.). Open University Press, Maidenhead.
- [9] K.-E. K. Bilstrup and M. H. Kaspersen. 2020. Staging reflections on ethical dilemmas in machine learning: A cardbased design workshop for high school students. In *Proceedings of the Designing Interactive Systems Conference*. ACM, New York.
- [10] P. Black, C. Harrison, C. Lee, B. Marshall, and W. Dylan. 2002. Working Inside the Black Box: Assessment for Learning in the Classroom. GLAssessm, London, UK.
- [11] Susanne Bødker. 2006. When second wave HCI meets third wave challenges. In Proceedings of the 4th Nordic Conference on Human–Computer Interaction: Changing Roles. Association for Computing Machinery, New York, NY, 1–8. DOI:https://doi.org/10.1145/1182475.1182476
- [12] Susanne Bødker and Morten Kyng. 2018. Participatory Design that Matters—Facing the Big Issues. ACM Transactions on Computer-Human Interaction 25, 1 (2018), 4:1-4:31. DOI: https://doi.org/10.1145/3152421
- [13] Wouter Boon and Ellen Moors. 2008. Exploring emerging technologies using metaphors A study of orphan drugs and pharmacogenomics. *Social Science and Medicine* 66, 9 (2008), 1915–1927. DOI:https://doi.org/10.1016/j. socscimed.2008.01.012
- [14] J. Bressler and J. Mohnke. 2015. School laboratories to teach robotics, smart home and artificial intelligence : From theory to practice. In *Proceedings of the 9th International Conference on Society and Information Technologies*). Retrieved February 26, 2022 from https://www.semanticscholar.org/paper/School-Laboratories-to-teach-Robotics-%2C-Smart-Home-Telematics/d638502a6be8dd3eaac153b92267852b4b40e9f3.
- [15] Harald Burgsteiner, Martin Kandlhofer, and Gerald Steinbauer. 2016. IRobot: Teaching the basics of artificial intelligence in high schools. *Proceedings of the AAAI Conference on Artificial Intelligence* 30, 1 (2016). Retrieved February 26, 2022 from https://ojs.aaai.org/index.php/AAAI/article/view/9864.
- [16] T. Burns and F. Gottschalk. 2019. Educating 21st Century Children: Emotional Well-being in the Digital Age. OECD Publishing, Paris. Retrieved from https://doi.org/10.1787/b7f33425-en
- [17] Jake Rowan Byrne, Katriona O'Sullivan, and Kevin Sullivan. 2017. An IoT and wearable technology hackathon for promoting careers in computer science. *IEEE Transactions on Education* 60, 1 (2017), 50–58. DOI:https://doi.org/10. 1109/TE.2016.2626252
- [18] Michelle Carney, Barron Webster, Irene Alvarado, Kyle Phillips, Noura Howell, Jordan Griffith, Jonas Jongejan, Amit Pitaru, and Alexander Chen. 2020. Teachable machine: Approachable web-based tool for exploring machine learning classification. In *Proceedings of the Extended Abstracts of the 2020 CHI Conference on Human Factors in Computing Systems*. Association for Computing Machinery, New York, NY, 1–8. DOI: https://doi.org/10.1145/3334480.3382839
- [19] S. Carretero, R. Vuorikari, and Y. Punie. 2017. DigComp 2.1: The Digital Competence Framework for Citizens with Eight Proficiency Levels and Examples of Use. Publications Office of the European Union, Luxembourg. Retrieved from DOI: 10.2760/38842
- [20] Michael E. Caspersen. 2022. Informatics as a fundamental discipline in general education: The danish perspective. In Proceedings of the Perspectives on Digital Humanism. Hannes Werthner, Erich Prem, Edward A. Lee and Carlo Ghezzi (eds.), Springer International Publishing, Cham, 191–200. DOI: https://doi.org/10.1007/978-3-030-86144-5_26
- [21] Michael E. Caspersen, Judith Gal-Ezer, Andrew McGettrick, and Enrico Nardelli. 2019. Informatics as a fundamental discipline for the 21st century. *Communications of the ACM* 62, 4 (2019), 58. DOI: https://doi.org/10.1145/3310330
- [22] Vicky Charisi, Laura Malinverni, Marie-Monique Schaper, and Elisa Rubegni. 2020. Creating opportunities for children's critical reflections on AI, robotics and other intelligent technologies. In *Proceedings of the 2020 ACM Interaction Design and Children Conference: Extended Abstracts.* Association for Computing Machinery, New York, NY, 89–95. DOI:https://doi.org/10.1145/3397617.3398063
- [23] Patricia Charlton and Katerina Avramides. 2016. Knowledge construction in computer science and engineering when learning through making. *IEEE Transactions on Learning Technologies* 9, 4 (2016), 379–390. DOI: https://doi.org/10. 1109/TLT.2016.2627567
- [24] S. Chaurdon, R. Di Gioia, and M. Gemo. 2017. Young Children (0-8) and Digital Technology A Qualitative Study Across Europe. Publications Office of the European Union, Luxembourg.
- [25] Jeannette Chin and Vic Callaghan. 2013. Educational living labs: A novel internet-of-things based approach to teaching and research. In Proceedings of the 2013 9th International Conference on Intelligent Environments. 92–99. DOI:https://doi.org/10.1109/IE.2013.48
- [26] Ban Hoe Chow. 2013. The virtual-reality campuses project. In Proceedings of the 3D Immersive and Interactive Learning. Yiyu Cai (ed.). Springer, Singapore, 59–70. DOI: https://doi.org/10.1007/978-981-4021-90-6_4
- [27] Bárbara Cleto, João Martinho Moura, Luís Ferreira, and Cristina Sylla. 2019. CodeCubes Playing with cubes and learning to code. In *Proceedings of the Interactivity, Game Creation, Design, Learning, and Innovation*. Springer International Publishing, Cham, 538–543. DOI: https://doi.org/10.1007/978-3-030-06134-0_58

- [28] Bárbara Cleto, Cristina Sylla, Luís Ferreira, and João Martinho Moura. 2020. "Play and Learn": Exploring Code-Cubes. In Proceedings of the Technology, Innovation, Entrepreneurship and Education. Springer International Publishing, Cham, 34–42. DOI: https://doi.org/10.1007/978-3-030-40180-1_4
- [29] Bárbara Cleto, Cristina Sylla, Luís Ferreira, and João Martinho Moura. 2020. CodeCubes: Coding with augmented reality. In Proceedings of the 1st International Computer Programming Education Conference. 9.
- [30] Nicoletta Corrocher, Franco Malerba, and Fabio Montobbio. 2003. The Emergence of New Technologies in the ICT Field: Main Actors, Geographical Distribution and Knowledge Sources. Department of Economics, University of Insubria. Retrieved February 26, 2022 from https://ideas.repec.org/p/ins/quaeco/qf0317.html.
- [31] Susan Cozzens, Sonia Gatchair, Jongseok Kang, Kyung-Sup Kim, Hyuck Jai Lee, Gonzalo Ordóñez, and Alan Porter. 2010. Emerging technologies: Quantitative identification and measurement. *Technology Analysis and Strategic Management* 22, 3 (2010), 361–376. DOI: https://doi.org/10.1080/09537321003647396
- [32] George S. Day and Paul J. H. Schoemaker. 2000. Avoiding the pitfalls of emerging technologies. California Management Review 42, 2 (2000), 8–33. DOI: https://doi.org/10.2307/41166030
- [33] Konstantinos T. Delistavrou and Achilles D. Kameas. 2017. Exploring ways to exploit UMI technologies in STEM education: Comparison of secondary computer science curricula of Greece, Cyprus and England. In Proceedings of the 2017 IEEE Global Engineering Education Conference. 1824–1830. DOI: https://doi.org/10.1109/EDUCON.2017.7943098
- [34] Xiaozhou Deng, Danli Wang, Qiao Jin, and Fang Sun. 2019. ARCat: A tangible programming tool for DFS algorithm teaching. In Proceedings of the 18th ACM International Conference on Interaction Design and Children. Association for Computing Machinery, New York, NY, 533–537. DOI: https://doi.org/10.1145/3311927.3325308
- [35] Manuj Dhariwal and Shruti Dhariwal. 2020. Let's Chance: Playful probabilistic programming for children. In Proceedings of the Extended Abstracts of the 2020 CHI Conference on Human Factors in Computing Systems. Association for Computing Machinery, New York, NY, 1–7. DOI: https://doi.org/10.1145/3334480.3383071
- [36] Shruti Dhariwal. 2018. Scratch memories: A visualization tool for children to celebrate and reflect on their creative trajectories. In *Proceedings of the 17th ACM Conference on Interaction Design and Children*. Association for Computing Machinery, Trondheim, Norway, 449–455. DOI: https://doi.org/10.1145/3202185.3202770
- [37] Mariette DiChristina and Bernard Meyerson. 2016. Top 10 Emerging Technologies of 2016 Insight Report. World Economic Forum. Retrieved February 26, 2022 from https://www3.weforum.org/docs/GAC16_Top10_Emerging_ Technologies_2016_report.pdf.
- [38] Mariette DiChristina and Bernard S. Meyerson. 2021. Top 10 Emerging Technologies 2021 Insight Report. World Economic Forum, Cologny/Geneva, Switzerland. Retrieved February 26, 2022 from https://www3.weforum.org/docs/ WEF_Top_10_Emerging_Technologies_of_2021.pdf.
- [39] Christian Dindler, Rachel Smith, and Ole Sejer Iversen. 2020. Computational empowerment: participatory design in education. CoDesign 16, 1 (2020), 66–80. DOI: https://doi.org/10.1080/15710882.2020.1722173
- [40] Monica Divitini, Michail N. Giannakos, Simone Mora, Sofia Papavlasopoulou, and Ole Sejer Iversen. 2017. Make2Learn with IoT: Engaging children into joyful design and making of interactive connected objects. In Proceedings of the 2017 Conference on Interaction Design and Children. Association for Computing Machinery, New York, NY, 757–760. DOI: https://doi.org/10.1145/3078072.3081312
- [41] Stefania Druga, Fee Christoph, and Amy J. Ko. 2022. Family as a Third Space for AI Literacies: How do children and parents learn about AI together? In *Proceedings of the 2022 CHI Conference on Human Factors in Computing Systems*. Association for Computing Machinery, New York, NY.
- [42] Stefania Druga, Sarah T. Vu, Eesh Likhith, and Tammy Qiu. 2019. Inclusive AI literacy for kids around the world. In Proceedings of FabLearn 2019. Association for Computing Machinery, New York, NY, 104–111. DOI: https://doi.org/ 10.1145/3311890.3311904
- [43] Steven D. Essinger and Gail L. Rosen. 2011. An introduction to machine learning for students in secondary education. In Proceedings of the 2011 Digital Signal Processing and Signal Processing Education Meeting. 243–248. DOI: https:// doi.org/10.1109/DSP-SPE.2011.5739219
- [44] Adson Marques da Silva Esteves, André Luiz Maciel Santana, and Rodrigo Lyra. 2019. Use of augmented reality for computational thinking stimulation through virtual. In *Proceedings of the 2019 21st Symposium on Virtual and Augmented Reality*. 102–106. DOI: https://doi.org/10.1109/SVR.2019.00031
- [45] Julián Estévez, X. Elizetxea, and Gorka Garate. 2019. Development of a workshop about artificial intelligence for schools. In Proceedings of the 13th International Conference on Soft Computing Models in Industrial and Environmental Applications. 530–535. DOI: https://doi.org/10.1007/978-3-319-94120-2_51
- [46] Julián Estévez, Gorka Garate, and Manuel Graña. 2019. Gentle introduction to artificial intelligence for high-school students using scratch. IEEE Access 7, (2019), 179027–179036. DOI: https://doi.org/10.1109/ACCESS.2019.2956136
- [47] European Commission. Digital Education Action Plan. (2021–2027). European Education Area: Quality Education and Training for all. Retrieved February 26, 2022 from https://education.ec.europa.eu/focus-topics/digital/educationaction-plan.

M. Van Mechelen et al.

- [48] Ignacio Evangelista, Germán Blesio, and Emanuel Benatti. 2018. Why Are We Not Teaching Machine Learning at High School? A proposal. In Proceedings of the 2018 World Engineering Education Forum - Global Engineering Deans Council. 1–6. DOI: https://doi.org/10.1109/WEEF-GEDC.2018.8629750
- [49] Christopher Frauenberger. 2019. Entanglement HCI the next wave? ACM Transactions on Computer-Human Interaction 27, 1 (2019), 2:1-2:27. DOI: https://doi.org/10.1145/3364998
- [50] Anna Fuste and Chris Schmandt. 2019. HyperCubes: A playful introduction to computational thinking in augmented reality. In Proceedings of the Extended Abstracts of the Annual Symposium on Computer-Human Interaction in Play Companion Extended Abstracts. Association for Computing Machinery, New York, NY, 379–387. DOI: https://doi.org/ 10.1145/3341215.3356264
- [51] George Gadanidis. 2017. Artificial intelligence, computational thinking, and mathematics education. The International Journal of Information and Learning Technology 34, 2 (2017), 133–139. DOI: https://doi.org/10.1108/IJILT-09-2016-0048
- [52] Juan David Rodríguez García, Jesús Moreno León, Marcos Román González, and Gregorio Robles. 2019. Developing computational thinking at school with machine learning: An exploration. In *Proceedings of the 2019 International Symposium on Computers in Education*. 1–6. DOI:https://doi.org/10.1109/SIIE48397.2019.8970124
- [53] Anna Gardeli and Spyros Vosinakis. 2019. ARQuest: A tangible augmented reality approach to developing computational thinking skills. In Proceedings of the 2019 11th International Conference on Virtual Worlds and Games for Serious Applications. 1–8. DOI: https://doi.org/10.1109/VS-Games.2019.8864603
- [54] Rosella Gennari, Alessandra Melonio, Mehdi Rizvi, and Andrea Bonani. 2017. Design of IoT tangibles for primary schools: A case study. In *Proceedings of the 12th Biannual Conference on Italian SIGCHI Chapter*. Association for Computing Machinery, New York, NY, 1–6. DOI: https://doi.org/10.1145/3125571.3125591
- [55] Michail Giannakos, Iro Voulgari, Sofia Papavlasopoulou, Zacharoula Papamitsiou, and Georgios Yannakakis. 2020. Games for artificial intelligence and machine learning education: Review and perspectives. In *Proceedings of the Non-Formal and Informal Science Learning in the ICT Era*. Michail Giannakos (ed.), Springer, Singapore, 117–133. DOI:https://doi.org/10.1007/978-981-15-6747-6_7
- [56] Dimitrios Glaroudis, Athanasios Iossifides, Natalia Spyropoulou, and Ioannis D. Zaharakis. 2018. Investigating secondary students' stance on IoT driven educational activities. In *Proceedings of the Ambient Intelligence* (Lecture Notes in Computer Science), Springer International Publishing, Cham, 188–203. DOI: https://doi.org/10.1007/978-3-030-03062-9_15
- [57] Dimitrios Glaroudis, Athanasios Iossifides, Natalia Spyropoulou, Ioannis D. Zaharakis, and Achilles D. Kameas. 2019. STEM learning and career orientation via IoT Hands-on activities in secondary education. In Proceedings of the 2019 IEEE International Conference on Pervasive Computing and Communications Workshops. 480–485. DOI:https:// doi.org/10.1109/PERCOMW.2019.8730759
- [58] Xiaoyan Gong, Yilin Wu, Zifan Ye, and Xiwei Liu. 2018. Artificial intelligence course design: iSTREAM-based visual cognitive smart vehicles. In Proceedings of the 2018 IEEE Intelligent Vehicles Symposium. 1731–1735. DOI: https://doi. org/10.1109/IVS.2018.8500457
- [59] Xiaoyan Gong, Liang Zhao, Ren Hao Tang, Ya Na Guo, Xiwei Liu, Jian He, Fei Yue Wang, Ying Tang, Wanruo Shi, Xiaojie Niu, and Xinzhu Wang. 2019. AI education system for primary and secondary schools. In *Proceedings of the ASEE Annual Conference and Exposition, Conference Proceedings.* Retrieved February 26, 2022 from https://www. researchwithrowan.com/en/publications/ai-education-system-for-primary-and-secondary-schools-2.
- [60] C. Goumopoulos, Olga Fragou, Nikolaos Chanos, Konstantinos Delistavrou, I. Zaharakis, Vasilis Stefanis, and A. Kameas. 2018. The UMI-Sci-Ed platform: Integrating UMI technologies to promote science education. In Proceedings of the CSEDU. DOI: https://doi.org/10.5220/0006686200780090
- [61] Sidhant Goyal, Rohan S. Vijay, Charu Monga, and Pratul Kalita. 2016. Code Bits: An inexpensive tangible computational thinking toolkit For K-12 curriculum. In *Proceedings of the TEI '16: 10th International Conference on Tangible, Embedded, and Embodied Interaction.* Association for Computing Machinery, New York, NY, 441–447. DOI:https://doi.org/10.1145/2839462.2856541
- [62] Maria J. Grant and Andrew Booth. 2009. A typology of reviews: An analysis of 14 review types and associated methodologies. *Health Information and Libraries Journal* 26, 2 (2009), 91–108. DOI:https://doi.org/10.1111/j.1471-1842.2009.00848.x
- [63] Polina Häfner, Victor Häfner, and Jivka Ovtcharova. 2014. Experiencing physical and technical phenomena in schools using virtual reality driving simulator. In *Proceedings of the Learning and Collaboration Technologies*. *Technology-Rich Environments for Learning and Collaboration* (Lecture Notes in Computer Science). Springer International Publishing, Cham, 50–61. DOI: https://doi.org/10.1007/978-3-319-07485-6_6
- [64] Mohanad Halaweh. 2013. Emerging technology: What is it? Journal of Technology Management and Innovation 8, 3 (2013), 108–115. DOI: https://doi.org/10.4067/S0718-27242013000400010

Emerging Technologies in K-12 Education: A Future HCI Research Agenda

- [65] C. Heinze, Janet Haase, and Helen Higgins. 2010. An action research report from a multi-year approach to teaching artificial intelligence at the K-6 level. In *Proceedings of the 24th AAAI Conference on Artificial Intelligence*. Retrieved February 26, 2022 from https://www.semanticscholar.org/paper/An-Action-Research-Report-from-a-Multi-Year-to-at-Heinze-Haase/a61a80d6a397c16e0f32f845a23b57e97e9d1bf5.
- [66] Tom Hitron, Yoav Orlev, Iddo Wald, Ariel Shamir, Hadas Erel, and Oren Zuckerman. 2019. Can children understand machine learning concepts? The Effect of Uncovering Black Boxes. In Proceedings of the 2019 CHI Conference on Human Factors in Computing Systems. Association for Computing Machinery, New York, NY, 1–11. DOI: https://doi. org/10.1145/3290605.3300645
- [67] Tom Hitron, Iddo Wald, Hadas Erel, and Oren Zuckerman. 2018. Introducing children to machine learning concepts through hands-on experience. In *Proceedings of the 17th ACM Conference on Interaction Design and Children*. Association for Computing Machinery, New York, NY, 563–568. DOI: https://doi.org/10.1145/3202185.3210776
- [68] Peter Holowka. 2020. TEACHING ROBOTICS DURING COVID-19: MACHINE LEARNING, SIMULATION, AND AWS DEEPRACER. 227–234. Retrieved February 26, 2022 from http://www.iadisportal.org/digital-library/teachingrobotics-during-covid-19-machine-learning-simulation-and-aws-deepracer.
- [69] Informatics for All. 2022. Mission. *Informatics for All.* Retrieved February 26, 2022 from https://www.informaticsforall.org/mission/#.
- [70] Qiao Jin, Danli Wang, Xiaozhou Deng, Nan Zheng, and Steve Chiu. 2018. AR-maze: a tangible programming tool for children based on AR technology. In *Proceedings of the 17th ACM Conference on Interaction Design and Children*. Association for Computing Machinery, New York, NY, 611–616. DOI: https://doi.org/10.1145/3202185.3210784
- [71] Sarah Judd. 2020. Activities for building understanding: How AI4ALL teaches AI to diverse high school students. In Proceedings of the 51st ACM Technical Symposium on Computer Science Education. Association for Computing Machinery, New York, NY, 633–634. Retrieved February 26, 2022 from https://doi.org/10.1145/3328778.3366990
- [72] R. Juskalian. 2020. 10 Breakthrough Technologies 2020. MIT Technology Review. Retrieved from https://www. technologyreview.com/10-breakthrough-technologies/2020/.
- [73] K. M. Kahn and N. Winters. 2017. Child-friendly Programming Interfaces to AI Cloud Services. Springer, Cham. Retrieved February 26, 2022 from https://ora.ox.ac.uk/objects/uuid:12124254-acce-4c11-a540-19e74530798d.
- [74] Ken Kahn, R. Megasari, E. Piantari, and E. Junaeti. 2018. AI programming by children using snap! Block programming in a developing country. In *Proceedings of the EC-TEL*.
- [75] Martin Kandlhofer, Gerald Steinbauer, Sabine Hirschmugl-Gaisch, and Petra Huber. 2016. Artificial intelligence and computer science in education: From kindergarten to university. In *Proceedings of the 2016 IEEE Frontiers in Education Conference*. 1–9. DOI: https://doi.org/10.1109/FIE.2016.7757570
- [76] Magnus Hoeholt Kaspersen, Karl-Emil Kjaer Bilstrup, Maarten Van Mechelen, Arthur Hjorth, Niels Olof Bouvin, and Marianne Graves Petersen. 2021. VotestratesML: A high school learning tool for exploring machine learning and its societal implications. In Proceedings of the FabLearn Europe /MakeEd 2021 - An International Conference on Computing, Design and Making in Education. Association for Computing Machinery, New York, NY, 1–10. DOI: https:// doi.org/10.1145/3466725.3466728
- [77] Kyung-Rok Kim, Seok-Hwan Jeong, Woo-Yong Kim, Youngjun Jeon, Kyung-Soo Kim, and Je-Hoon Hong. 2017. Design of small mobile robot remotely controlled by an Android operating system via bluetooth and NFC communication. In Proceedings of the 2017 14th International Conference on Ubiquitous Robots and Ambient Intelligence. 913–915. DOI: https://doi.org/10.1109/URAI.2017.7992864
- [78] Eric Klopfer and Josh Sheldon. 2010. Augmenting your own reality: Student authoring of science-based augmented reality games. New Directions for Youth Development 128 (2010), 85–94. DOI: https://doi.org/10.1002/yd.378
- [79] Marge Kusmin, Merike Saar, and Mart Laanpere. 2018. Smart schoolhouse designing IoT study kits for projectbased learning in STEM subjects. In *Proceedings of the 2018 IEEE Global Engineering Education Conference*. 1514–1517. DOI:https://doi.org/10.1109/EDUCON.2018.8363412
- [80] Phoebe Lin and Jessica Van Brummelen. 2021. Engaging teachers to Co-Design integrated AI curriculum for K-12 classrooms. In *Proceedings of the 2021 CHI Conference on Human Factors in Computing Systems*. Association for Computing Machinery, New York, NY, 1–12. DOI: https://doi.org/10.1145/3411764.3445377
- [81] Annabel Lindner, Stefan Seegerer, and Ralf Romeike. 2019. Unplugged activities in the context of AI. In Proceedings of the Informatics in Schools. New Ideas in School Informatics (Lecture Notes in Computer Science). Springer International Publishing, Cham, 123–135. DOI: https://doi.org/10.1007/978-3-030-33759-9_10
- [82] Breanne K. Litts, Apoorva Chauhan, Chase K. Mortensen, and Kamaehu Matthias. 2019. I'm drowning in squirrels! How children embody and debug computational algorithms through designing mixed reality games. In *Proceedings* of the 18th ACM International Conference on Interaction Design and Children. Association for Computing Machinery, New York, NY, 267–273. DOI: https://doi.org/10.1145/3311927.3323129
- [83] Sonia Livingstone, Monica Bulger, Patrick Burton, Emma Day, Eva Lievens, Ingrida Milkaite, Tom De Leyn, Marijn Martens, Ricarose Roque, Katharine Sarikakis, Mariya Stoilova, and Ralf De Wolf. 2022. Children's privacy and

digital literacy across cultures: Implications for education and regulation. In *Proceedings of the Learning to Live with Datafication*. Routledge.

- [84] Sonia Livingstone and Amanda Third. 2017. Children and young people's rights in the digital age: An emerging agenda. New Media and Society 19, 5 (2017), 657–670. DOI: https://doi.org/10.1177/1461444816686318
- [85] Duri Long and Brian Magerko. 2020. What is AI Literacy? Competencies and design considerations. In Proceedings of the 2020 CHI Conference on Human Factors in Computing Systems, Association for Computing Machinery, New York, NY, 1–16. Retrieved February 26, 2022 from https://doi.org/10.1145/3313831.3376727
- [86] Maria Luce Lupetti. 2017. Shybo. An open-source low-anthropomorphic robot for children. HardwareX 2, (2017), 50–60. DOI: https://doi.org/10.1016/j.ohx.2017.08.003
- [87] Stéphane Magnenat, Morderchai Ben-Ari, Severin Klinger, and Robert W. Sumner. 2015. Enhancing robot programming with visual feedback and augmented reality. In *Proceedings of the 2015 ACM Conference on Innovation* and Technology in Computer Science Education. Association for Computing Machinery, New York, NY, 153–158. DOI:https://doi.org/10.1145/2729094.2742585
- [88] Radu Mariescu-Istodor and Ilkka Jormanainen. 2019. Machine learning for high school students. In Proceedings of the 19th Koli Calling International Conference on Computing Education Research. Association for Computing Machinery, New York, NY, 1–9. DOI: https://doi.org/10.1145/3364510.3364520
- [89] Ben R. Martin. 1995. Foresight in science and technology. Technology Analysis & Strategic Management 7, 2 (1995), 139–168. DOI: https://doi.org/10.1080/09537329508524202
- [90] Giovanna Mascheroni and Donell Holloway (Eds.). 2019. The Internet of Toys: Practices, Affordances and the Political Economy of Children's Smart Play. Palgrave Macmillan.
- [91] Juan Mateu and Xavier Alamán. 2012. An experience of using virtual worlds and tangible interfaces for teaching computer science. In *Proceedings of the Ubiquitous Computing and Ambient Intelligence* (Lecture Notes in Computer Science). Springer, Berlin, 478–485. DOI: https://doi.org/10.1007/978-3-642-35377-2_66
- [92] Juan Mateu and Xavier Alamán. 2013. CUBICA: An example of mixed reality. JUCS Journal of Universal Computer Science 19, 17 (2013), 2598–2616. DOI: https://doi.org/10.3217/jucs-019-17-2598
- [93] Anna Mavroudi, Monica Divitini, Francesco Gianni, Simone Mora, and Dag R. Kvittem. 2018. Designing IoT applications in lower secondary schools. In *Proceedings of the 2018 IEEE Global Engineering Education Conference*. 1120–1126. DOI: https://doi.org/10.1109/EDUCON.2018.8363355
- [94] Jenny Millea, Ian Green, and Garry Putland. 2005. Emerging Technologies: A Framework for Thinking. ACT Department of Education and Training, Adelaide, Australia. Retrieved February 26, 2022 from http://hdl.voced.edu.au/ 10707/54114
- [95] Atherton Mook. 2019. Virtual reality based environment for immersive, hands-on learning. In Proceedings of the Practice and Experience in Advanced Research Computing on Rise of the Machines (learning). Association for Computing Machinery, New York, NY, 1–4. DOI: https://doi.org/10.1145/3332186.3338099
- [96] Filipe T. Moreira, Mário Vairinhos, and Fernando Ramos. 2020. Open IoT technologies in the classroom a case study on the student's perception. In Proceedings of the 2020 15th Iberian Conference on Information Systems and Technologies. 1–6. DOI: https://doi.org/10.23919/CISTI49556.2020.9141106
- [97] Ana Mouta, Eva Torrecilla Sánchez, and Ana Pinto Llorente. 2019. Blending machines, learning, and ethics. In Proceedings of the 7th International Conference on Technological Ecosystems for Enhancing Multiculturality. Association for Computing Machinery, New York, NY, 993–998. DOI: https://doi.org/10.1145/3362789.3362909
- [98] Georgios Mylonas, Dimitrios Amaxilatis, Lidia Pocero, Iraklis Markelis, Joerg Hofstaetter, and Pavlos Koulouris. 2018. Using an educational IoT Lab Kit and gamification for energy awareness in european schools. In *Proceedings* of the Conference on Creativity and Making in Education. Association for Computing Machinery, Trondheim, Norway, 30–36. DOI: https://doi.org/10.1145/3213818.3213823
- [99] Georgios Mylonas, Dimitrios Amaxilatis, Lidia Pocero, Iraklis Markelis, Joerg Hofstaetter, and Pavlos Koulouris. 2019. An educational IoT lab kit and tools for energy awareness in European schools. *International Journal of Childcomputer Interaction* 20, (2019), 43–53. DOI: https://doi.org/10.1016/j.ijcci.2019.03.003
- [100] Taro Narahara and Yoshihiro Kobayashi. 2018. Personalizing homemade bots with plug & play AI for STEAM education. In *Proceedings of the SIGGRAPH Asia 2018 Technical Briefs*. Association for Computing Machinery, New York, NY, 1–4. DOI: https://doi.org/10.1145/3283254.3283270
- [101] National Science and Technology Council. 2016. Preparing for the Future of Artificial Intelligence. US Government -Executive Office of the President.
- [102] Davy Tsz Kit Ng, Jac Ka Lok Leung, Kai Wah Samuel Chu, and Maggie Shen Qiao. 2021. AI Literacy: Definition, teaching, evaluation and ethical issues. Proceedings of the Association for Information Science and Technology 58, 1 (2021), 504–509. DOI:https://doi.org/10.1002/pra2.487
- [103] Simone Opel, Michael Schlichtig, and Carsten Schulte. 2019. Developing teaching materials on artificial intelligence by using a simulation game (Work in Progress). In *Proceedings of the 14th Workshop in Primary and Secondary*

Computing Education. Association for Computing Machinery, New York, NY, 1-2. DOI: https://doi.org/10.1145/ 3361721.3362109

- [104] Elisaweta Ossovski and Michael Brinkmeier. 2019. Machine learning unplugged development and evaluation of a workshop about machine learning. In *Proceedings of the Informatics in Schools. New Ideas in School Informatics*. Springer International Publishing, Cham, 136–146. DOI: https://doi.org/10.1007/978-3-030-33759-9_11
- [105] Koki Ota, Tsuyoshi Nakajima, and Hiroki Suda. 2020. A Short-Term course of STEAM education through IoT exercises for high school students. In Proceedings of the 2020 IEEE 44th Annual Computers, Software, and Applications Conference. 153–157. DOI: https://doi.org/10.1109/COMPSAC48688.2020.00028
- [106] Matthew J. Page, Joanne E. McKenzie, Patrick M. Bossuyt, Isabelle Boutron, Tammy C. Hoffmann, Cynthia D. Mulrow, Larissa Shamseer, Jennifer M. Tetzlaff, Elie A. Akl, Sue E. Brennan, Roger Chou, Julie Glanville, Jeremy M. Grimshaw, Asbjørn Hróbjartsson, Manoj M. Lalu, Tianjing Li, Elizabeth W. Loder, Evan Mayo-Wilson, Steve McDonald, Luke A. McGuinness, Lesley A. Stewart, James Thomas, Andrea C. Tricco, Vivian A. Welch, Penny Whiting, and David Moher. 2021. The PRISMA 2020 statement: an updated guideline for reporting systematic reviews. *BM*7 372, (2021), n71. DOI:https://doi.org/10.1136/bmj.n71.
- [107] Seymour A. Papert. 1980. Mindstorms: Children, Computers, And Powerful Ideas (1st. ed.). Basic Books, New York.
- [108] Seymour Papert and Idit Harel. 1991. Situating constructionism. *Constructionism* 36, 2 (1991), 1–11.
- [109] Judy Perry. 2019. Gaming the schoolyard: Promoting high school students' collaborative learning through geolocative mobile game design. In *Proceedings of the CSCL*. DOI: https://doi.org/10.22318/CSCL2019.957
- [110] Mark Petticrew and Helen Roberts. 2006. Systematic Reviews in the Social Sciences: A Practical Guide. Wiley-Blackwell, New Jersey.
- [111] Prapa Rattadilok, Chris Roadknight, and Li Li. 2018. Teaching students about machine learning through a gamified approach. In Proceedings of the 2018 IEEE International Conference on Teaching, Assessment, and Learning for Engineering. 1011–1015. DOI: https://doi.org/10.1109/TALE.2018.8615279
- [112] Antonio Rizzo, Francesco Montefoschi, Maurizio Caporali, Giovanni Burresi, Antonio Rizzo, Francesco Montefoschi, Maurizio Caporali, and Giovanni Burresi. 2018. UAPPI: A platform for extending app inventor towards the worlds of IoT and machine learning. In *Proceedings of the Innovative Methods, User-Friendly Tools, Coding, and Design Approaches in People-Oriented Programming*. Steve Goschnick (ed.), IGI Global, 88–109. Retrieved from https://doi.org/10.4018/978-1-5225-5969-6.ch003
- [113] Juan David Rodríguez-García, Jesús Moreno-León, Marcos Román-González, and Gregorio Robles. 2020. Introducing artificial intelligence fundamentals with learningML: Artificial intelligence made easy. In Proceedings of the 8th International Conference on Technological Ecosystems for Enhancing Multiculturality. Association for Computing Machinery, New York, NY, 18–20. DOI: https://doi.org/10.1145/3434780.3436705
- [114] Daniele Rotolo, Diana Hicks, and Ben R. Martin. 2015. What is an emerging technology? *Research Policy* 44, 10 (2015), 1827–1843. DOI: https://doi.org/10.1016/j.respol.2015.06.006
- [115] Alpay Sabuncuoglu. 2020. Designing one year curriculum to teach artificial intelligence for middle school. In Proceedings of the 2020 ACM Conference on Innovation and Technology in Computer Science Education. Association for Computing Machinery, New York, NY, 96–102. DOI: https://doi.org/10.1145/3341525.3387364
- [116] Wasin Sae-jeng, Sunti Tuntrakool, Somchai Maunsaiyat, and Piya Supavarasuwat. 2020. AWC 708C laser cutting manual learning set. In Proceedings of the 2020 59th Annual Conference of the Society of Instrument and Control Engineers of Japan. 1763–1769. DOI: https://doi.org/10.23919/SICE48898.2020.9240436
- [117] Bawornsak Sakulkueakulsuk, Siyada Witoon, Potiwat Ngarmkajornwiwat, Pornpen Pataranutaporn, Werasak Surareungchai, Pat Pataranutaporn, and Pakpoom Subsoontorn. 2018. Kids making AI: Integrating machine learning, gamification, and social context in STEM education. In Proceedings of the 2018 IEEE International Conference on Teaching, Assessment, and Learning for Engineering. 1005–1010. DOI: https://doi.org/10.1109/TALE.2018.8615249
- [118] Pratiti Sarkar, Kapil Kadam, and Jayesh S. Pillai. 2019. Collaborative approaches to Problem-Solving on lines and angles using augmented reality. In Proceedings of the 2019 IEEE Tenth International Conference on Technology for Education. 193–200. DOI: https://doi.org/10.1109/T4E.2019.00-24
- [119] Pratiti Sarkar, Kapil Kadam, and Jayesh S. Pillai. 2020. Learners' approaches, motivation and patterns of problemsolving on lines and angles in geometry using augmented reality. *Smart Learning Environments* 7, 1 (2020), 17. DOI:https://doi.org/10.1186/s40561-020-00124-9
- [120] Marie-Monique Schaper, Laura Malinverni, and cristina valero. 2020. Robot Presidents: Who should rule the world? Teaching critical thinking in AI through reflections upon food traditions. In *Proceedings of the 11th Nordic Conference on Human–computer Interaction: Shaping Experiences, Shaping Society.* Association for Computing Machinery, New York, NY, 1–4. Retrieved February 26, 2022 from https://doi.org/10.1145/3419249.3420085
- [121] Marie-Monique Schaper, Rachel Charlotte Smith, Mariana Aki Tamashiro, Maarten Van Mechelen, Mille Skovhus Lunding, Karl-Emil Kjær Bilstrup, Magnus Høholt Kaspersen, Kasper Løvborg Jensen, Marianne Graves Petersen, and Ole Sejer Iversen. 2022. Principles for Teenagers' Learning About Emerging Technologies and Their Societal

Impact: Machine Learning and Augmented Reality in K-12 Education. Social Science Research Network, Rochester, NY. DOI: https://doi.org/10.2139/ssrn.4013380

- [122] Alexander Scheidt and Tim Pulver. 2019. Any-Cubes: A children's toy for learning AI: Enhanced play with deep learning and MQTT. In *Proceedings of the Mensch und Computer 2019*. Association for Computing Machinery, New York, NY, 893–895. DOI: https://doi.org/10.1145/3340764.3345375
- [123] Gelson Schneider, Flavia Bernardini, and Clodis Boscarioli. 2020. Teaching CT through internet of things in high school: Possibilities and reflections. In Proceedings of the 2020 IEEE Frontiers in Education Conference. 1–8. DOI: https://doi.org/10.1109/FIE44824.2020.9274184
- [124] Ricky Setiawan, Adnan Rachmat Anom Besari, Iwan Kurnianto Wibowo, Muh Rifqi Rizqullah, and Dias Agata. 2018. Mobile visual programming apps for internet of things applications based on raspberry Pi 3 platform. In Proceedings of the 2018 International Electronics Symposium on Knowledge Creation and Intelligent Computing. 199–204. DOI: https://doi.org/10.1109/KCIC.2018.8628483
- [125] Vandit Sharma, Jeevankur Talukdar, and Kaushal Bhagat. 2019. CodAR: An augmented reality based game to teach programming. In Proceedings of the 27th International Conference on Computers in Education.
- [126] Nicole Sintov, Debarun Kar, Thanh Nguyen, Fei Fang, Kevin Hoffman, Arnaud Lyet, and Milind Tambe. 2016. From the lab to the classroom and beyond: Extending a game-based research platform for teaching AI to diverse audiences. In Proceedings of the 13th AAAI Conference on Artificial Intelligence. AAAI Press, Phoenix, Arizona, 4107–4112.
- [127] David Smahel, Hana MacHackova, Giovanna Mascheroni, Lenka Dedkova, Elisabeth Staksrud, Kjartan Olafsson, Sonia Livingstone, and Uwe Hasebrink. 2020. EU Kids Online 2020: Survey Results from 19 Countries. London School of Economics and Political Science, London, UK. Retrieved February 26, 2022 from https://doi.org/10.21953/lse. 47fdeqj01ofo
- [128] Hannah Snyder. 2019. Literature review as a research methodology: An overview and guidelines. Journal of Business Research 104, (2019), 333–339. DOI: https://doi.org/10.1016/j.jbusres.2019.07.039
- [129] Natalia Spyropoulou, Dimitrios Glaroudis, Athanasios Iossifides, and Ioannis D. Zaharakis. 2020. Fostering secondary students' STEM career awareness through IoT hands-On educational activities: Experiences and lessons learned. *IEEE Communications Magazine* 58, 2 (2020), 86–92. DOI: https://doi.org/10.1109/MCOM.001.1900288
- [130] Raji Srinivasan. 2008. Sources, characteristics and effects of emerging technologies: Research opportunities in innovation. *Industrial Marketing Management* 37, 6 (2008), 633–640. DOI: https://doi.org/10.1016/j.indmarman.2007.12. 003
- [131] Bernd Carsten Stahl. 2011. What Does the Future Hold? A critical view of emerging information and communication technologies and their social consequences. In *Proceedings of the Researching the Future in Information Systems* (IFIP Advances in Information and Communication Technology), Springer, Berlin, 59–76. DOI:https://doi.org/10.1007/ 978-3-642-21364-9_5
- [132] Anselm Strauss and Juliet Corbin. 1998. Basics of Qualitative Research: Techniques and Procedures for Developing Grounded Theory, (2nd. ed.). Sage Publications, Inc, Thousand Oaks, CA.
- [133] Mike Tissenbaum, Mark A. Sherman, Joshua Sheldon, and Hal Abelson. 2018. From Computational Thinking to Computational Action: Understanding Changes in Computational Identity through App Inventor and the Internet of Things. (July 2018). Retrieved February 26, 2022 from https://repository.isls.org//handle/1/788
- [134] Tapani Toivonen, Ilkka Jormanainen, Juho Kahila, Matti Tedre, Teemu Valtonen, and Henriikka Vartiainen. 2020. Co-Designing machine learning apps in K-12 with primary school children. In Proceedings of the 2020 IEEE 20th International Conference on Advanced Learning Technologies. 308–310. DOI: https://doi.org/10.1109/ICALT49669.2020.00099
- [135] Richard J. Torraco. 2005. Writing integrative literature reviews: Guidelines and examples. Human Resource Development Review 4, 3 (2005), 356–367. DOI: https://doi.org/10.1177/1534484305278283
- [136] David Touretzky, Christina Gardner-McCune, Cynthia Breazeal, Fred Martin, and Deborah Seehorn. 2019. A year in K-12 AI education. AI Magazine 40, 4 (2019), 88–90. DOI: https://doi.org/10.1609/aimag.v40i4.5289
- [137] David Touretzky, Christina Gardner-McCune, Fred Martin, and Deborah Seehorn. 2019. Envisioning AI for K-12: What should every child know about AI? *Proceedings of the AAAI Conference on Artificial Intelligence* 33, 01 (2019), 9795–9799. DOI: https://doi.org/10.1609/aaai.v33i01.33019795
- [138] Andrea C. Tricco, Erin Lillie, Wasifa Zarin, Kelly K. O'Brien, Heather Colquhoun, Danielle Levac, David Moher, Micah D. J. Peters, Tanya Horsley, Laura Weeks, Susanne Hempel, Elie A. Akl, Christine Chang, Jessie McGowan, Lesley Stewart, Lisa Hartling, Adrian Aldcroft, Michael G. Wilson, Chantelle Garritty, Simon Lewin, Christina M. Godfrey, Marilyn T. Macdonald, Etienne V. Langlois, Karla Soares-Weiser, Jo Moriarty, Tammy Clifford, Özge Tunçalp, and Sharon E. Straus. 2018. PRISMA extension for scoping reviews (PRISMA-ScR): Checklist and explanation. Annals of Internal Medicine 169, 7 (2018), 467–473. DOI: https://doi.org/10.7326/M18-0850
- [139] UNICEF. 2019. Workshop Report: AI and Child Rights Policy. United Nations Children's Fund (UNICEF), New York, NY.

Emerging Technologies in K-12 Education: A Future HCI Research Agenda

- [140] United States Government. National Artificial Intelligence Initiative: Overseeing and Implementing the United States National AI Strategy. AI GOV. Retrieved February 26, 2022 from https://www.ai.gov/.
- [141] Alex Vakaloudis, Kieran Delaney, Brian Cahill, and Jacqueline Kehoe. 2019. Enabling primary school teachers to deliver STEM programmes with the internet of things: Challenges and recipes for success. In Proceedings of the 2019 11th International Conference on Education Technology and Computers. Association for Computing Machinery, New York, NY, 80–83. DOI: https://doi.org/10.1145/3369255.3369312
- [142] Teemu Valtonen, Matti Tedre, Kati Mäkitalo, and Henriikka Vartiainen. 2019. Media literacy education in the age of machine learning. *Journal of Media Literacy Education* 11, 2 (2019), 20–36. DOI: https://doi.org/10.23860/JMLE-2019-11-2-2
- [143] Jessica Van Brummelen, Viktoriya Tabunshchyk, and Tommy Heng. 2021. Alexa, Can I Program You?: Student perceptions of conversational artificial intelligence before and after programming alexa. In *Proceedings of the Interaction Design and Children*. Association for Computing Machinery, New York, NY, 305–313. DOI: https://doi.org/10.1145/ 3459990.3460730
- [144] Maarten Van Mechelen, Line Have Musaeus, Ole Sejer Iversen, Christian Dindler, and Arthur Hjorth. 2021. A systematic review of empowerment in child-computer interaction research. In *Proceedings of the International Conference* on Interaction Design and Children. Association for Computing Machinery, Athens, Greece.
- [145] Maarten Van Mechelen, Marie-Louise Wagner, Gökçe Elif Baykal, Rachel Charlotte Smith, and Ole Sejer Iversen. 2021. Digital design literacy in K-9 education: Experiences from pioneer teachers. In *Proceedings of the International Conference on Interaction Design and Children*. Association for Computing Machinery, Athens, Greece.
- [146] Henriikka Vartiainen, Matti Tedre, and Teemu Valtonen. 2020. Learning machine learning with very young children: Who is teaching whom? International Journal of Child–Computer Interaction 25, (2020), 100182. DOI: https://doi.org/ 10.1016/j.ijcci.2020.100182
- [147] Henriikka Vartiainen, Tapani Toivonen, Ilkka Jormanainen, Juho Kahila, Matti Tedre, and Teemu Valtonen. 2020. Machine learning for middle-schoolers: Children as designers of machine-learning apps. In Proceedings of the 2020 IEEE Frontiers in Education Conference. 1–9. DOI: https://doi.org/10.1109/FIE44824.2020.9273981
- [148] Anu Vazhayil, Radhika Shetty, Rao R. Bhavani, and Nagarajan Akshay. 2019. Focusing on teacher education to introduce AI in schools: Perspectives and illustrative findings. In *Proceedings of the 2019 IEEE 10th International Conference on Technology for Education*. 71–77. DOI: https://doi.org/10.1109/T4E.2019.00021
- [149] Igor Verner, Dan Cuperman, Tal Romm, Michael Reitman, Shi Kai Chong, and Zoe Gong. 2020. Intelligent robotics in high school: An educational paradigm for the industry 4.0 era. In *Proceedings of the 21st International Conference* on Interactive Collaborative Learning. 824–832. DOI: https://doi.org/10.1007/978-3-030-11932-4_76
- [150] Igor Verner, Michael Reitman, Dan Cuperman, Toria Yan, Eldad Finkelstein, and Tal Romm. 2019. Exposing robot learning to students in augmented reality experience. In *Proceedings of the Smart Industry and Smart Education* (Lecture Notes in Networks and Systems). Springer International Publishing, Cham, 610–619. DOI:https://doi.org/ 10.1007/978-3-319-95678-7_67
- [151] Olga Viberg and Anna Mavroudi. 2018. The role of ubiquitous computing and the internet of things for developing 21st century skills among learners: Experts' views. In *Proceedings of the Lifelong Technology-Enhanced Learning* (Lecture Notes in Computer Science). Springer International Publishing, Cham, 640–643. DOI:https://doi.org/10. 1007/978-3-319-98572-5_63
- [152] Christiane Gresse Von Wangenheim, Nathalia da Cruz Alves, Marcelo F. Rauber, Jean C. R. Hauck, and Ibrahim H. Yeter. 2021. A proposal for performance-based assessment of the learning of machine learning concepts and practices in K-12. *Informatics in Education* (2021). DOI: https://doi.org/10.15388/infedu.2022.18
- [153] Mihaela Vorvoreanu, Colin M. Gray, Paul Parsons, and Nancy Rasche. 2017. Advancing UX education: A model for integrated studio pedagogy. In *Proceedings of the 2017 CHI Conference on Human Factors in Computing Systems*. Association for Computing Machinery, New York, NY, 1441–1446. DOI:https://doi.org/10.1145/3025453.3025726
- [154] Iro Voulgari, Marvin Zammit, Elias Stouraitis, Antonios Liapis, and Georgios Yannakakis. 2021. Learn to machine learn: Designing a game based approach for teaching machine learning to primary and secondary education students. In *Proceedings of the Interaction Design and Children*. Association for Computing Machinery, New York, NY, 593–598. DOI: https://doi.org/10.1145/3459990.3465176
- [155] Xiaoyu Wan, Xiaofei Zhou, Zaiqiao Ye, Chase K. Mortensen, and Zhen Bai. 2020. SmileyCluster: Supporting accessible machine learning in K-12 scientific discovery. In *Proceedings of the Interaction Design and Children Conference*. Association for Computing Machinery, New York, NY, 23–35. DOI: https://doi.org/10.1145/3392063.3394440
- [156] Lauren Wilcox, Betsy DiSalvo, Dick Henneman, and Qiaosi Wang. 2019. Design in the HCI classroom: Setting a research agenda. In Proceedings of the 2019 on Designing Interactive Systems Conference. Association for Computing Machinery, New York, NY, 871–883. DOI: https://doi.org/10.1145/3322276.3322381
- [157] Randi Williams, Hae Won Park, and Cynthia Breazeal. 2019. A is for artificial intelligence: The impact of artificial intelligence activities on young children's perceptions of robots. In Proceedings of the 2019 CHI Conference on Human

Factors in Computing Systems. Association for Computing Machinery, New York, NY, 1–11. DOI: https://doi.org/10. 1145/3290605.3300677

- [158] Randi Williams, Hae Won Park, Lauren Oh, and Cynthia Breazeal. 2019. PopBots: Designing an artificial intelligence curriculum for early childhood education. *Proceedings of the AAAI Conference on Artificial Intelligence* 33, 01 (2019), 9729–9736. DOI: https://doi.org/10.1609/aaai.v33i01.33019729
- [159] , Xiaojuan Ma, and John Huen. 2019. When schools meet artificial intelligence in Hong Kong. ACM Inroads 10, 4 (2019), 43–46. DOI: https://doi.org/10.1145/3369739
- [160] Benjamin Xie, Erik Harpstead, Betsy DiSalvo, Petr Slovak, Ahmed Kharrufa, Michael J. Lee, Viktoria Pammer-Schindler, Amy Ogan, and Joseph Jay Williams. 2019. Learning, Education, and HCI. In Proceedings of the Extended Abstracts of the 2019 CHI Conference on Human Factors in Computing Systems. Association for Computing Machinery, New York, NY, 1–4. DOI: https://doi.org/10.1145/3290607.3311761
- [161] D. Yamada-Rice, F. Mushtaq, A. Woodgate, D. Bosmans, A. Douthwaite, I. Douthwaite, W. Harris, R. Holt, D. Kleeman, J. Marsh, E. Milovidov, M. Mon Williams, B. Parry, A. Riddler, P. Robinson, D. Rodrigues, S. Thompson, and S. Whitley. 2017. *Children and Virtual Reality: Emerging Possibilities and Challenges*. Retrieved February 26, 2022 from http://digilitey.eu
- [162] Yanfang Yu and Yuan Chen. 2018. Design and development of high school artificial intelligence textbook based on computational thinking. Open Access Library Journal 5, 9 (2018), 1–15. DOI: https://doi.org/10.4236/oalib.1104898
- [163] Ioannis D. Zaharakis, Nicolas Sklavos, and Achilles Kameas. 2016. Exploiting ubiquitous computing, mobile computing and the internet of things to promote science education. In *Proceedings of the 2016 8th IFIP International Conference on New Technologies, Mobility and Security*. 1–2. DOI: https://doi.org/10.1109/NTMS.2016.7792451
- [164] Bieke Zaman, Maarten Van Mechelen, and Lizzy Bleumers. 2018. When toys come to life: Considering the internet of toys from an animistic design perspective. In *Proceedings of the 17th ACM Conference on Interaction Design and Children*. Association for Computing Machinery, New York, NY, 170–180. DOI: https://doi.org/10.1145/3202185.3202745
- [165] Daniel Zeng. 2013. From computational thinking to AI thinking [A letter from the editor]. IEEE Intelligent Systems 28, 6 (2013), 2–4. DOI: https://doi.org/10.1109/MIS.2013.141
- [166] Abigail Zimmermann-Niefield, Shawn Polson, Celeste Moreno, and R. Benjamin Shapiro. 2020. Youth making machine learning models for gesture-controlled interactive media. In *Proceedings of the Interaction Design and Children Conference*. Association for Computing Machinery, New York, NY, 63–74. DOI: https://doi.org/10.1145/3392063. 3394438
- [167] Abigail Zimmermann-Niefield, Makenna Turner, Bridget Murphy, Shaun K. Kane, and R. Benjamin Shapiro. 2019. Youth learning machine learning through building models of athletic moves. In *Proceedings of the 18th ACM International Conference on Interaction Design and Children*. Association for Computing Machinery, Boise, ID, 121–132. DOI: https://doi.org/10.1145/3311927.3323139

Received 1 March 2022; revised 27 June 2022; accepted 8 August 2022

47:40